STUDENTS' COUNCIL

Tuesday February 28, 2012 Council Chambers 2-1 University Hall

ORDER PAPER (SC 2011-20)

2011-20/1	SPEAKER'S BUSINESS
2011-20/1a	Announcements – The next meeting of Students' Council will take place on Tuesday, March 13, 2012
2011-20/2	PRESENTATIONS
2011-20/3	EXECUTIVE COMMITTEE REPORT
2011-20/3a	Executive Committee Report
	Please see document SC 11-20.01
2011-20/4	BOARD AND COMMITTEE REPORTS
2011-20/4a	Ruling 2011-05 of the DIE Board (VP External Re: Bylaw 2200)
	Please see document SC 11-20.02
2011-20/5	QUESTION PERIOD
2011-20/6	BOARD AND COMMITTEE BUSINESS
2011-20/6a	KHINDAYAMAGISHI/POWER MOVE THAT upon the recommendation of the Policy Committee Students' Council approve the Food Policy in first reading based on the following principles
	Please see document SC 11-18.03
2011-20/6b	KHINDA/CRONE MOVE THAT upon the recommendation of the Policy Committee Students' Council approve the Health and Wellness in first reading based on the following principles
	Please see document SC 11-18.04
2011-20/60	KHINDA/ZINYEMBA MOVE THAT upon the recommendation of the Policy Committee Students' Council approve the Smoking Policy first reading based on the following principles

Please see document SC 11-18.05

2011-20/6d CRONE/CHEEMA MOVE THAT Students' Council approve the 2012-2013

Budget Principles on the recommendation of the Budget and Finance

Committee.

Please see document SC 11-18.06

2011-20/7 GENERAL ORDERS

2011-20/7a KHINDA MOVES TO appoint one (1) member of Students' Council to the

Policy Committee

2011-20/7b TIGHE/CSORBA MOVE TO authorize the President and General Manager to

sign the final agreement for the Physical Activity and Wellness Centre

Please see document SC 11-20.07

2011-20/8 INFORMATION ITEMS

2011-20/8a Policy Committee- Summary report to Council

Please see document SC 11-20.08

2011-20/8b BFC- Summary Report to Council

Please see document SC 11-20.09

2011-20/8c Farid Iskandar, VP External- Report

Please see document SC 11-20.10

2011-20/8d Colten Yamagishi, VP Student Life- Report

Please see document SC 11-20.11

2011-20/8e Votes and Proceedings

Please see document SC 11-20.12

January 31

CHEEMA/YAMAGISHI MOVED THAT the Executive Committee approve a projects allocation not to exceed \$5000.00 for the Vice President External Farid Iskandar, President Rory Tighe and Director of Research and Political Affairs Justin Williams for the purpose of attending the CASA AGM from March 13-16, 2012 in Halifax, Nova Scotia.

5/0/0 CARRIED

CHEEMA/TIGHE MOVED THAT the Executive Committee approve a contingency reserve expense not to exceed \$2500.00 for the purpose of purchasing a new freezer for Juicy.

5/0/0 CARRIED

February 2

CHEEMA/ISKANDAR MOVED THAT the Executive Committee approve the Orientation Logo.

5/0/0 CARRIED

February 10

YAMAGISHI/TIGHE MOVED THAT the Executive Committee approve a projects allocation not to exceed \$900.00 for the purpose of food for Lobby Training.

5/0/0 CARRIED

February 14

ISKANDAR/YAMAGISHI MOVED THAT the Executive Committee approve a Projects Allocation not to exceed \$1050.00 for the purpose of sponsoring Health Week.

5/0/0 CARRIED

ISKANDAR/YAMAGISHI MOVED THAT the Executive Committee approve a Projects Allocation not to exceed \$1450.00 for the purpose of sending the Senior Manager Programming & Venues, Jennifer Wanke to the NACA National Conference.

3/2/0 CARRIED

TIGHE/CSORBA MOVED THAT the Executive Committee approve the new Discover Governance Program logo.

5/0/0 CARRIED

Style of Cause: Reference Re Bylaw 2200

Hearing Number: 05 - 2011/2012

Hearing Date: February 12th 2012

DIE Board Panel Members: Kathleen Elhatton-Lake, Associate Chief Tribune, Chair;

Rebecca Gormley, Tribune;

Eve Coppinger, Tribune

Appearing for the Applicant: Farid Iskandr, Vice-President External University of Alberta

Students' Union

Intervener: Zach Fentiman Chief Returning Officer

Background:

- [1] Farid Iskandr, VP External of the Students' Council requested that the DIE board provide an interpretation of §12(2) of Bylaw 2000. Mr. Iskandr notes that §12(2) requires that a member of Students' Council who is running for election must take a leave of absence from their duties for the period beginning with the nomination deadline and ending with the closing of voting for the election they are running in. He also stated that his concern was that many of these positions could not nominate a proxy.
- [2] He also notes that many of the executive positions of the Students' Council are cross appointed to other committees and boards. Of particular note in this case are the appointments to the General Faculties Council, the University of Alberta Senate and the University of Alberta Board of Governors.
- [3] Bylaw 100 §20 lays out the nominations to the General Faculties Council, the University of Alberta Senate and the University of Alberta Board of Governors. The CRO notes that these are nominations and require acceptance before they are formally members of their respective boards. Additionally both the CRO and the VP External stress that these boards have meetings during election season and the student representation on these boards is of particular importance. Specifically this year the University of Alberta Senate is searching for a new Chancellor. To facilitate this search they have struck a Search Committee for the Chancellor. This committee required to have a Students' Union member on it. Farid Iskandr was nominated for the position and accepted for the position on May 17th 2011. Mr. Iskandr is also the Students' Council Vice-President External and as such sits on the University of Alberta Senate. There is no requirement in Students' Union Bylaws that the Students' Union must nominate a Student Senator or the

Vice-President External for the position on the Search Committee which is struck every four years.

Issues:

- [4] Is a member of the Students' Council, who is taking a leave of absence as per Bylaw 2200 Section 12(2), required to take leave from their duties on the:
 - a) University of Alberta Board of Governors;
 - b) University of Alberta General Faculties Council;
 - c) University of Alberta Senate?
- [5] Is a member of the Students' Council allowed to choose a proxy to represent them while they are taking a leave of absence in order to run for re-election?

Relevant Legislation:

- [6] Excerpts from Bylaw 100
- 20. Nominations to University Bodies
- (1) The four individuals nominated by the Students' Union to the University of Alberta Senate shall be
- (a) the Students' Union Vice President (External), and
- (b) three members of the Students' Union selected by Students' Council, after receipt by the latter of a recommendation from the Council Administration Committee.
- (2) The two individuals nominated as ex-officio representatives of the Students' Union to the University of Alberta General Faculties Council shall be
- (a) the Students' Union Vice President (Academic), and
- (b) the Students' Union Vice President (Student Life).
- (3) The two individuals nominated by the Students' Union to the University of Alberta Board of Governors shall be
- (a) the Students' Union President, and
- (b) the Undergraduate Board of Governors Representative.
- (4) The Undergraduate Board of Governors Representative shall be elected at the same time and in the same fashion as the Students' Union President and Vice Presidents

- [7] Excerpts from Bylaw 2000
- 12. Restrictions on Candidate Nominees
- (1) No member shall be nominated for more than one (1) of the positions contested in each election.
- (2) Members of Students' Council and its standing committees, in order for their nomination papers to be valid are required to take a leave of absence from their duties for the period beginning with the nomination deadline and ceasing with the conclusion of voting of the election in which they are contesting a position. The following exceptions apply:
- a. Any member of Students' Council contesting an executive position when the race is uncontested

For the purpose of this section, any race solely contested by a joke candidate shall be considered uncontested.

- [8] Excerpts from Bylaw 1100
- 4. (1) The President
- a. shall develop and maintain awareness of significant issues that impact students;
- b. is the primary spokesperson for the Students' Union;
- c. shall supervise the affairs of the Students' Union subject to the authority of the Executive Committee and Students' Council; and
- d. has the authority to direct the General Manager.
- (2) The Vice President (External)
- a. shall develop and maintain awareness of significant issues external to the University of Alberta that impact students;
- b. shall advocate for the Students' Union to the federal, provincial, and municipal governments; and
- c. shall lead and coordinate the Students' Union's efforts on advocacy external to the University of Alberta.
- (3) The Vice President (Academic)
- a. shall develop and maintain awareness of significant academic issues that impact students;
- b. shall advocate on behalf of the Students' Union on academic issues; and

- c. shall lead and coordinate the Students' Union's efforts on academic advocacy.
- (4) The Vice President (Operations and Finance)
- a. shall develop and maintain awareness of the Students' Union's operations and finances;
- b. shall oversee those portions of the Students' Union's budget process falling within the realm of the Executive Committee;
- c. shall provide operational and financial advice to the Executive Committee and its members; and
- d. shall, in the absence of the President,
- i. chair the Executive Committee, and
 - ii. have the authority to direct the General Manager.
- (5) The Vice President (Student Life)
- a. shall develop and maintain awareness of significant non-academic issues internal to the University of Alberta that impact students;
- b. shall advocate on behalf of the Students' Union on non-academic issues internal to the University of Alberta; and
- c. shall lead and coordinate the Students' Union's efforts on advocacy on non-academic issues internal to the University of Alberta.

Decision

- [9] In cases where the individual hold another position solely due to their elected Students' Union position then that position will be considered a duty of the elected position and a leave of absence will be required from it as well. If the position is one acquired separately from the Student's Union position then the individual will not be required to take a leave of absence. If bylaw requires that an elected member of Students Council be the nominee for a position then that position will be considered a duty.
- [10] A elected member of Students' Council who is taking a leave of absence in order to run for election as per §12(2) of Bylaw 2000 may not appoint a proxy for their positions.

Analysis:

[11] First, DIE board must examine what constitutes a duty as per §12(2) of Bylaw 2000. Bylaw 1100 lists what each position shall do. These are clearly duties. However, they are not the limit on what constitutes a duty. The list is short and clearly meant as an outline rather than a

specific list. DIE board also turns to Bylaw 100(20) to see the nominations of the Students' Union.

- [12] Here we see that some positions are automatically nominated to other student bodies such as the GFC, Board of Directors and Senate. Though these are nominations and not appointments of the Students Council the nominations process is not a competitive process with other nominees and the individual is the automatic nominee of the Students' Council by virtue of their Students' Union position. This would apply to the positions that the President and Undergraduate Board of Governors Representative hold on the University of Alberta Board of Governors, the Vice-President Academic and Vice-President Student Life on the General Faculties Council and the Vice-President External on the Senate. Unlike the Senate nominees other than the Vice-President External the Undergraduate Board of Governors Representative is an elected position that is simultaneously a voting Students' Council position.
- [13] If any members of the Students' Council are serving on committees that are part of these bodies in capacities that are not due to their positions they would not be required to take a leave of absence from those separate positions. If a member of Students' Council is holding a position that any student would be able to run or apply for and receive without being a member of Students' Council then they may continue serving in that capacity. It is separate from their Students' Council position and if it is not related should not be considered a duty of a position under §12(2) of Bylaw 2000.
- [14] DIE board also was asked to clarify if an individual taking a leave of absence from a position due to election was able to appoint a proxy to act for them in their other positions. DIE board rules that an individual taking a leave of absence due to running for re-election is unable to appoint their own proxy. In examining the purpose of By-Law 2200 Section 12(2) it is clear that the purpose is to prevent a Students' Union member from being able to gain an unfair advantage from currently occupying their position. As such, allowing them to appoint their own proxy would fundamentally undermine this purpose.

Food Policy - Third Draft

WHEREAS access to good preparation resources on campus is currently limited;

WHEREAS nutrition plays a vital role in the health and wellness of students and therefore can have a significant impact on their academic success;

WHEREAS access to a variety of healthy food and beverage options available on campus is currently limited;

WHEREAS food and beverages sold on campus do not meet the needs of students with a variety of dietary restrictions;

WHEREAS nutritional information provided for food products sold on campus is currently limited;

WHEREAS food and beverages incur a significant cost to students;

WHEREAS students living in Lister Centre are required to purchase a mandatory meal plan that can only be used to purchase food and beverages from specific food vendors on campus who accept ONEcards as payment;

WHEREAS food and beverages have a social and environmental impact;

BE IT RESOLVED THAT the Students' Union advocates that the University increases the availability of diverse and healthy food options;

BE IT FURTHER RESOLVED THAT the Students' Union advocates that the University provides an increase of affordable food and beverage options;

BE IT FURTHER RESOLVED THAT the Student's Union advocates that the University provides and improves access to food preparation resources;

BE IT FURTHER RESOLVED THAT the Students' Union advocates that the University provides and improves access to water fountains and water bottle refilling stations;

BE IT FURTHER RESOLVED THAT the Students' Union advocates that the University encourages the sale of organic, local, or Fair Trade food and beverage options when competitively priced to reduce its adverse ecological and social impact;

BE IT FURTHER RESOLVED THAT the Students' Union advocates that the University expands its ONEcard service to as many food vendors as possible, as to provide more choice for students who wish to eat on campus.

BE IT FURTHER RESOLVED THAT the Students' Union advocates that the University provide easily accessible nutritional information for all of its food products;

YAMAGISHIKHINDA/POWER move that the Policy Committee approve the Food Policy in first reading based on the following principles:

- 1. The Students' Union advocates that the University increases the availability of diverse and healthy food options and healthy alternatives;
- 2. The Students' Union advocates that the University provides an increase of affordable food<u>and beverage</u> options;
- 3. The Student's Union advocates that the University provides and improves access to food preparation resources;
- 4. The Students' Union advocates that the University provides and improves access to water fountains and water bottle refilling stations;
- 5. The Students' Union advocates that the University encourages the sale of organic¹, local, or fair Fair trade Trade certified food and beverage options when competitively priced to reduce our its adverse ecological and social impact;
- 6. The Students' Union advocates that the University reduces the ONEcard vendor levy to promote the expansion of its ONEcard service to food-vendors providing more choice for students who wish to eat on campus;
- 7. The Students' Union advocates that the University provides easily accessible nutritional information for all of its food and beverage products; and
- 8. That this policy will expire April 30th 2013.

http://www.ifoam.org/growing organic/3 advocacy lobbying/eng leaflet PDF/Climate C hange.pdf (2007).

^{1 &}quot;Organic agriculture is a production system that sustains the health of soils, ecosystems and people. It relies on ecological processes, biodiversity and cycles adapted to local conditions, rather than the use of inputs with adverse effects. Organic agriculture combines tradition, innovation and science to benefit the shared environment and promote fair relationships and a good quality of life for all involved." International Federation of Organic Agriculture Movements (IFOAM). "Organic Agriculture's Role in Countering Climate Change."

HEALTH AND WELLNESS POLICY

WHEREAS the health and wellness of an individual encompasses emotional, mental, physical, spiritual, and sexual health;

WHEREAS the health and wellness of students may be affected by the physical environment on the University of Alberta Campus

WHEREAS the health and wellness of a student has a direct effect on academic and social success:

WHEREAS students are increasingly succeptible to high levels of stress due to heavy workloads;

WHEREAS there are limited fitness and physical wellness facilities on campus;

WHEREAS studies show that students exhibit low rates of sexually healthy behaviour;

WHEREAS there is demand for increased quantity and quality of interfaith prayer and meditation space on campus;

WHEREAS high risk alcohol consumption is prevalent within the student population; and

WHEREAS addictive behavior, including, but not limited to, addiction to computer/internet use and to cognitive-enhancing drugs can be damaging to a student's academic success:

BE IT RESOLVED THAT the Students' Union advocates that the University provide an increase in mental health resources, with an emphasis on counseling services;

BE IT RESOLVED THAT the Students' Union advocates that the University provide an increase in physical wellness space;

BE IT RESOLVED THAT the Students' Union advocates that the University provide an increase in sexual health awareness and education;

BE IT RESOLVED THAT the Students' Union advocates that the University provides an increase to interfaith prayer, meditation, and spiritual space on campus; and

BE IT RESOLVED THAT the Students' Union advocates that the University provide resources that foster a culture of moderation in regards to alcohol consumption.

Smoking on Campus SMOKING POLICY

WHEREAS The Students' Union is committed to the health and wellness of all its employees and its members;

WHEREAS The Students' Union is committed to initiatives promoting sustainability on campus

WHEREAS The University of Alberta is a world leading institution in health and cancer research;

WHEREAS Environmental tobacco smoke (ETS) is a known carcinogen, and poses a serious preventable health risk

WHEREAS cigarette butts and ETS pose a considerable hazard to the local ecosystem

WHEREAS the reduction of cigarette butts and litter on campus has the potential to reduce maintenance and disposal costs;

WHEREAS the Government of Alberta does not allow smoking: in public buildings, on children's playgrounds, on patios or porches of restaurants, or within 5m of any entrance to such buildings;

WHEREAS the University of Alberta defined campuses are: Campus Saint Jean, Enterprise Square, North Campus, South Campus, and Augustana;

WHEREAS a campus-wide ban would have a high potential of placing members of the campus community in risky situations;

BE IT RESOLVED THAT the Students' Union advocate that the University of Alberta create a policy that would limitrestrict smoking on all of its' campuses to appropriate designated sites moking areas which minimize the campus populations exposure to ETS.

n	D)								
$\mathbf{\sigma}$	т	$\overline{}$	т	т	т	т	т	т	т	ī

BE IT RESOLVED THAT the Students' Union advocate that the University of Alberta create a policy that would ban smoking on all of its' campuses.

Q30. Which of the following would you be most supportive of in terms of restrictions on smoking on campus?

Count Percent

3426	47.22%	A campus wide smoking ban (no smoking anywhere on campus, indoors or outdoors)
2485	34.25%	Smoking be restricted to designated areas only
1345	18.54%	No change to the status quo
7256	Respondents	



February 15, 2012

To: Students' Council

Re: Proposed Budget Priorities of Students' Council, University of Alberta Students' Union for the Fiscal Year Beginning May 1, 2012 and Ending April 30, 2013

This document is intended to serve as an outline of the fiscal priorities of the Students' Union, as per the legislative process for the budget, outlined in the Standing Orders of Students' Council.

1. General

The four key focuses of the Students' Union are the following:

- Governance, representation, and advocacy,
- Providing student services,
- Operating the Students' Union Building, and
- Operating businesses for the purpose of funding Students' Union activities.

Total operating and capital expenditures will not exceed the amount of total operating and fee revenues. The only exception to this would be council approved capital purchases that do not cause a deficit on an accrual basis.

All departments must work within the mandate of the Students' Union and activities or operations deemed to be beyond its scope shall be eliminated. The Students' Union Strategic Plan 2011-2014 shall be used as a guide to assess relevance and value prior to such decisions.

External entities such as dedicated fee units shall be managed as per existing Students' Union bylaw and any applicable contracts.

Capital expenditures shall be incurred based upon the following criteria:

- Replacement due to wear and tear.
- To ensure continued operations of a particular unit.
- To enhance the security or functionality of a particular unit.
- To strengthen the viability of a particular unit.

Where possible, precedence in capital budgeting shall be given to expenditures contemplated in the 5-Year Capital Plan.

In the event that one section of this document conflicts with any other section, the Budget and Finance Committee shall identify such a conflict and propose any necessary changes in the presentation of the final budget.



Unless otherwise noted operating and fixed costs increase at a rate, relative to the previous fiscal year, not higher than inflation for the province of Alberta and activity of the Students' Union will continue in a manner consistent with the precedent set in previous fiscal years.

The Students' Union shall not budget for a deficit. Growth in a department shall be offset by strengthened revenue streams in that department or sector, where possible.

The Students' Union shall implement all necessary financial restraints and audits as required by the Post-Secondary Learning Act and its own bylaws.

2. Staff

Staff costs shall change as specified in the CUPE collective agreement. Most managerial and other non-unionized staff are entitled to a merit increase of up to 4%, as per their respective employment agreements.

3. Fees

Operating and dedicated fees shall increase in accordance with Students' Union Bylaw 3000. The consumer price index shall be calculated as 2.9%.

4. Advocacy and Representation

The Advocacy and Representation units of the Students' Union are the following:

- The Executive Committee, along with each individual executive member,
- Students' Council,
- The Department of Research & Political Affairs,
- Governance,
- Elections,
- CAUS, and
- CASA.

The Students' Union shall allocate resources to provide sufficient staffing to support student governance objectives.

5. Services

The service units of the Students' Union are the following:

- Infolink,
- Centre for Student Development,
- Ombudservice,
- Student Financial Aid and Information Centre,
- Student Group Services,



- Safewalk,
- Peer Support Centre,
- Handbook, and
- SUSTAIN SU.

The Students' Union shall allocate additional resources to support the development of information and communication infrastructure.

6. Business

The business units of the Students' Union are the following:

- SUBmart,
- SUBtitles,
- SUBprint,
- Cram Dunk,
- Canada Post (Postal Outlet),
- Room at the Top,
- Dewey's,
- L'Express and L'Express Catering,
- Juicy, and
- SUB Rental Operations.

The Students' Union business units shall not budget for a deficit prior to cost apportionment allocations. All reasonable attempts to maximize the profitability of these business units, consistent with the mandate of the Students' Union as a service provider, shall be made.

An average margin increase to non-academic materials not resulting from decreasing costs shall not exceed 10%.

Where costs have increased in academic material areas, there shall be a margin increase to compensate.

The Students' Union shall allocate additional resources to upgrade preparation facilities and equipment in Food & Beverage units.

The Students' Union shall reduce costs associated with SUBtitles by adjusting operating hours to reflect seasonal demand.

7. Entertainment and Programming

The Entertainment and Programming units of the Students' Union are the following:



- Week of Welcome,
- Myer Horowitz Theatre,
- Dinwoodie Lounge,
- Antifreeze,
- SUB Programming, and
- Alternative Entertainment and Programming.

For the purpose of the budget principles discussion, programming that takes place in a Students' Union venue or event that is not specifically mentioned in the above list shall be considered as separate from the entertainment and programming units.

Where possible, all entertainment and programming units should plan to recover all costs. Admission sales, sponsorship and donations are acceptable sources of revenue. Exceptions shall be made clear in the presentation of the final budget.

8. Administration, Operations, and Support

The administrative, operational, and support units of the Students' Union are:

- Marketing,
- Sponsorship,
- Facilities, and
- Administration.

Administration, operations, and support costs shall be apportioned to all departments of the Students' Union on a usage basis.

The Students' Union shall allocate additional resources toward future improvements and/or expansion of the Students' Union Building.

9. Changes from 11/12 Budget Principles

The consumer price index shall be calculated as 2.9%.

Staff costs shall change as specified in the CUPE collective agreement. Most managerial and other non-unionized staff are entitled to a merit increase of up to 4%, as per their respective employment agreements.

The Students' Union shall allocate additional resources to support the development of information and communication infrastructure.

The Students' Union shall reduce costs associated with SUBtitles by adjusting operating hours to reflect seasonal demand.



This document is intended to serve as an outline of the fiscal priorities of the Students' Union, as per the legislative process for the budget, outlined in the Standing Orders of Students' Council.

The Students' Union shall allocate additional resources to upgrade preparation facilities and equipment in Food & Beverage units.

The Students' Union Strategic Plan 2011-2014 shall be used as a guide to assess relevance and value prior to such decisions.

Where possible, precedence in capital budgeting shall be given to expenditures contemplated in the 5-Year Capital Plan.

Growth in a department shall be offset by strengthened revenue streams in that department or sector, where possible.

Awards is now consolidated under the Student Financial Aid & Information Centre and the Students' Union shall allocate resources to provide sufficient staffing to support the Awards program within the Student Financial Aid and Information Centre *has been removed*, as this was completed in the 11/12 budget.

The Environmental Coordination Office of Students has been renamed to SUSTAIN SU.

Orientation and Transition Programs (Centre for Student Development) has been renamed to Centre for Student Development

The Students' Union shall allocate additional resources to enhance product development expertise in Food & Beverage units has been removed, as this was completed in the 11/12 budget.

Additional resources shall be allocation for capital improvements for SFAIC and the Peer Support Centre *has been removed*, as this was completed in the 11/12 budget.

¹ Executive compensation for the 2011/2012 year was increased to \$33,000 per executive. See Nov 30/10 SC motion 6(a). *has been removed*, as this was completed in the 11/12 budget.

THIS AGREEMENT	dated effective this	day of _	, 2012

BETWEEN:

THE GOVERNORS OF THE UNIVERSITY OF ALBERTA (the "University")

-and-

THE STUDENTS' UNION OF THE UNIVERSITY OF ALBERTA (the "Students' Union")

-and-

THE GRADUATE STUDENTS' ASSOCIATION OF THE UNIVERSITY OF ALBERTA (the "GSA")

PHYSICAL ACTIVITY AND WELLNESS (PAW) CENTRE PROJECT AGREEMENT

WHEREAS:

- **A.** The Parties share a mission to promote the holistic wellness and sustainability of the campus community, through the provision of space, facilities, and programs that provide for physical, social, and associated activities, including larger and more modern fitness facilities (the "Shared Mission");
- **B.** The Parties have agreed to the construction of the PAW Centre and the renovation of the Other Renovated Areas in order to further the Shared Mission; and
- C. The Parties wish to set out the principles governing the establishment of the PAW Centre and renovation of the Other Renovated Areas, and their respective obligations and contributions to the planning, design, construction/renovation, commissioning, funding, maintenance and ongoing operations of the PAW Centre and the Other Renovated Areas.
- **D.** The following principles (collectively being the "Agreement Principles") are key in the development and implementation of this Agreement:
 - i. University decision making is inherently based on a collegial relationship;
 - **ii.** When meetings are conducted between presidents of the Students' Union and GSA and senior administration of the University, Parties are treated as equals at the table;

- iii. There exists equality and respect for the University as an institution, and what it is about;
- iv. The University resolves disputes internally whenever possible;
- v. There exists a right of recourse on decisions to the Provost and Vice-President (Academic), as set out in further detail in this Agreement;
- vi. This Agreement has been drafted by building upon the principles set out in the Referenda passed by the memberships of the Associations and attached in Schedule A-3; it is the parties' intention that this Agreement be consistent with, and provide further context and details in relation to, those principles;
- vii. Revenue derived from any operation of, and delivery of programs within, the PAW Centre and Other Renovated Areas by the University over and above operating costs or costs of delivering the program, respectively (and for greater clarity, rental fees payable by Campus Recreation in the normal course of operations to another unit of the University shall be included in the calculation of such costs), shall be dedicated to the payment of operating and capital repair/replacement costs of the PAW Centre and Other Renovated Areas during the Term hereof;
- viii. No portion of the revenue generated from the Fee, nor the Student Contributions Loan, nor any interest earned or accruing on such funds, may be used to fund any portion of the Project consisting of the Academic and Research Areas;
 - From and after the Acceptance Date, no modifications of the Student Focused Areas resulting in a substantial reduction of usable area or a substantial reduction of functionality of an area shall be made without the agreement of the Associations, other than in accordance with section 11.01 or section 12 of this Agreement; and
 - x. The primary focus of the operation of the Project is to provide health, activity and wellness services to Students that are of the highest quality possible, subject always to the efficient and prudent financial operations of the Project.

NOW THEREFORE in consideration of the above, the payment of \$1.00 by the Associations to the University and by the University to the Associations and other good and valuable consideration, the receipt, sufficiency and benefit of which is hereby acknowledged by each of the Parties, each of the Parties agrees with each of the other Parties as follows:

INTERPRETATION

1.01 Definitions

The definitions set forth in **Schedule A-1 (Definitions)** to this Agreement shall govern the meaning of all defined terms used in this Agreement, unless there is something in the subject matter or context that is expressly inconsistent therewith.

1.02 Construction and Interpretation

In this Agreement, including the recitals, Schedules and appendices to this Agreement, except where expressly stated to the contrary or the context otherwise requires, the construction and interpretation of provisions shall be in accordance with **Schedule A-2 (Interpretation and General Contractual Terms)**.

1.03 Schedules

All Schedules are included in and form part of this Agreement. The following schedules form part of this Agreement:

Schedule A-1 – Definitions

Schedule A-2 – Interpretation and General Contractual Terms

Schedule A-3 – Referendum Questions

Schedule B-1 – Functional Requirements

Schedule C-1 – Operational Requirements

Schedule D-1 – Form of Lease Agreement

Schedule E-1 – Committee Procedures

Schedule F-1 – Conflict and Dispute Resolution Procedure

1.04 Conflicting Provisions

If there is any conflict between:

- a. the Agreement Principles and any portion of the main body of this Agreement and any of the Schedules hereto, the Agreement Principles shall prevail;
- b. the main body of this Agreement and any of the Schedules hereto other than Schedule B-1 (Functional Requirements), Schedule C-1 (Operational Requirements) or Schedule D-1 (Form of Lease Agreement), the provisions of the main body of this Agreement shall prevail;
- c. the main body of this Agreement and Schedule B-1 (Functional Requirements), Schedule C-1 (Operational Requirements) or Schedule D-1 (Form of Lease Agreement), the provisions of Schedule B-1 (Functional Requirements), Schedule C-1 (Operational Requirements), or Schedule D-1 (Form of Lease Agreement) as the case may be, shall prevail;
- d. Schedule B-1 (Functional Requirements), and Schedule C-1 (Operational Requirements), Schedule B-1 (Functional Requirements) shall prevail;

e. Schedule D-1 (Form of Lease Agreement) and either Schedule B-1 (Functional Requirements) or Schedule C-1 (Operational Requirements), Schedule D-1 (Form of Lease Agreement) shall prevail.

1.05 Purpose of the Project

The purpose of the Project is to support the Shared Mission through the creation of a unified health, wellness and fitness complex – a focal point – that integrates disparate parts of existing buildings with a clear circulation order as well as an improved security and control plan, the resulting comprehensive centre for sport, research, teaching, and other fitness and physical activities to strengthen the quality of the undergraduate and the graduate student experience (the "Project Purpose").

1.06 Cooperation

Without expanding or modifying the obligations of the Parties set out in this Agreement or creating additional obligations to those already set forth herein, the Parties acknowledge that this Agreement is intended to achieve efficiencies, allocate risks and provide benefits for each of the Parties and the ultimate service recipients from or through the PAW Centre and the Other Renovated Areas. Throughout the Term, the Parties will cooperate with each other with a view to furthering the purposes and objectives of this Agreement, including, without limitation, through the establishment of such committees, groups and lines of communication as may reasonably be requested by either Party.

1.07 No Joint Venture

No partnership, joint venture, employment, fiduciary or agency relationship is created by this Agreement or under this Agreement.

PLANNING, DESIGN, CONSTRUCTION/RENOVATION, COMMISSIONING AND RENOVATION

2.01 PAW Centre and Other Renovated Areas Commitment

The University shall arrange for the planning, design, construction/renovation, and commissioning of the PAW Centre and Other Renovated Areas, in particular performing the Work and Operation and Maintenance, upon and subject to the terms and conditions set forth in this Agreement and in the Lease Agreement. Without restricting the generality of the foregoing, except as otherwise set forth in this Agreement or in the Lease Agreement, the University is fully responsible for:

a. the design, construction/renovation and commissioning of all parts of the PAW Centre and Other Renovated Areas substantially in accordance with the plans and specifications in the Design Development Report as aspects of such Design Development Report are modified and subsequent plans and specifications are developed in accordance with **Schedule B-1 (Functional Requirements)**;

- **b.** providing or arranging for all financing of the Work;
- the correction of all and any deficiencies or defects (latent and patent) pertaining to the design and construction of the Project that materially and negatively affect the functionality of any of the Student Focused Areas;
- **d.** Operation and Maintenance of the PAW Centre and Other Renovated Areas during the Term in accordance with the Operational Requirements and this Agreement; and
- e. the compliance of the performance of the Work and Operation and Maintenance of the PAW Centre and Other Renovated Areas with all applicable laws.

2.02 Modifications to the Project or Work

The Project or the Work may be modified from time to time by the University as is necessary or desirable, provided such modification is made in accordance with **Schedule B-1 (Functional Requirements)**.

CONDITION PRECEDENT & ACCEPTANCE DATE

3.01 Conditions

This Agreement is subject to the following conditions precedent:

- a. The University shall apply and be approved for financing not exceeding THIRTY MILLION (\$30,000,000.00) DOLLARS, for an amortizing debenture or mortgage, with a term and amortization period not exceeding 35 years, an effective interest rate not exceeding 6.25%, and on terms otherwise satisfactory to the University in its sole discretion ("Student Contributions Loan"). The purpose of this financing is set out in s. 4.02 herein;
- b. The University shall obtain such other financing as it deems necessary, and shall obtain sufficient commitments from the Province of Alberta and other donors in order to enable the University to meet its financial obligations set out in this Agreement, on terms satisfactory to the University in its sole discretion; and
- c. The University shall receive all required approvals from the Government of Alberta pursuant to the *Post-Secondary Learning Act* (Alberta), as amended from time to time, for the borrowing described in sections 3.01(a) and (b);

on or before the 30th day of September, 2012 (the "Condition Day").

3.02 Satisfaction or Waiver

Unless otherwise agreed in writing, the conditions set out in section 3.01 are for the sole benefit of the University. The University shall use commercially reasonable efforts to satisfy the conditions set out in section 3.01. The University may unilaterally waive or confirm satisfaction of these conditions by giving a notice to such effect to the Students' Union and the GSA on or before the Condition Day. If such a notice has not been given on or before the Condition Day, then this Agreement is terminated, effective on the day following the Condition Day.

3.03 Notice of Satisfaction or Waiver

The University may give written notice to the Students' Union and the GSA on or before the Condition Day advising that a condition will not be waived, has not been satisfied and will not be satisfied on or before the Condition Day. If that notice is given, then this Agreement is ended upon the giving of that notice.

3.04 Acceptance Date

The Acceptance Date shall be deemed to be the first day of the month in which the University delivers to the Associations and to the Chair of the Steering Committee a certificate of the Consulting Architect for the Project certifying firstly that on or before the date of such certificate the Work has been substantially performed, as such term is used in the Builders' Lien Act (Alberta), and secondly that the Student Focused Areas are operational and ready for occupancy and use for their intended purpose. The lack of furniture, fixtures or equipment the acquisition or installation of which is the responsibility of the Associations shall not result in the Student Focused Areas not being operational and ready for occupancy and use for their intended purpose. The Steering Committee shall consider the certification that the Student Focused Areas are operational and ready for occupancy and use for their intended purpose and if either the President (or designate) of the Student's Union or the President (or designate) of the GSA advises the Steering Committee meeting at which such consideration takes place that such President's (or designate's) Association disagrees with such certification, or reasonably believes the Agreement Principles are not being complied with, such disagreement shall be treated as a conflict and shall be resolved in accordance with Schedule F-1 (Conflict and Dispute Resolution Procedure).

FINANCIAL CONTRIBUTIONS

4.01 Interim Financing of Allowable Student Costs

Prior to the Acceptance Date, the Allowable Student Costs shall be covered by bridge financing provided at the option of the University by either:

a. the University at a deemed rate of interest on Allowable Student Costs from the month in which the same are paid by the University until the repayment of the Allowable Student Costs from the proceeds of the Student Contributions Loan calculated and compounded monthly at an annual interest rate equal to the average 91 day treasury bill rate as posted by the Bank of Canada for that month; or

b. by third party lenders acting at arm's length, at an interest rate not to exceed 6.25% per annum;

(all deemed interest chargeable pursuant to section 4.01(a) and all interest, carrying costs and any other associated costs, fees, charges and expenses charged pursuant to section 4.01(b) being the "Bridge Financing Costs"). The total of Allowable Student Costs covered by bridge financing as contemplated within this Section shall not exceed Thirty Million Dollars (\$30,000,000). The bridge financing contemplated within this Section shall be paid out by the University utilizing the proceeds of the Student Contribution Loan as contemplated within this Agreement no later than six (6) months following the Acceptance Date.

4.02 Permitted Use of Student Contributions Loan

The University shall use the principal amount received pursuant to the Student Contributions Loan to repay the Allowable Student Costs and to the extent there is a remaining balance of funds available from the Student Contributions Loan to apply such balance to the Bridge Financing Costs and for no other purpose.

4.03 Maximum Principal

The principal amount of the Student Contributions Loan shall not exceed Thirty Million Dollars (\$30,000,000), or the total of Allowable Student Costs and Bridge Financing Costs, whichever is less.

4.04 Student Fee Funding

From and after the Acceptance Date, the University shall levy a fee on each undergraduate Student and each graduate Student, subject to the following terms:

- a. The maximum Fee collected from each undergraduate or graduate Student shall be twenty-nine (\$29.00) dollars for each of the Fall Term and the Winter Term, and fourteen dollars and fifty cents (\$14.50) for each of the Spring Term and the Summer Term:
- **b.** If the Acceptance Date occurs during any such term other than the first day of the first month of a Fall Term, Winter Term, Spring Term or Summer Term, subject always to the Agreement Principles:
 - the Fee for the portion of the term in which the Acceptance Date occurs shall be equal to the proportion that the remaining number of months of that term (calculated from the first day of the next month immediately after the month in which the Acceptance Date has occurred) to the end of that term, is of the total number of months of that term; and

- ii. the Fee for such portion of the term shall be assessed on the first day of the next month, immediately after the month in which the Acceptance Date has occurred;
- c. The Fee shall be applied and collected from Students other than Students who are only enrolled in courses designated as off campus or in courses at Augustana Campus, in accordance with University policy, as amended from time to time;
- **d.** Students who are only enrolled in courses designated as off campus or in courses at Augustana Campus may choose whether to pay the Fee;
- e. Students who elect to pay the Fee as provided in section 4.04(c) will have equal access rights to those set out in section 7.03 hereto;
- **f.** Subject to the foregoing, the University will determine the period for which the Fee shall be assessed in its discretion, except that
 - i. such period shall not exceed thirty-five (35) years, commencing upon the date of the first assessment of the Fee by the University;
 - ii. and in any event shall not extend longer than is necessary in order to reimburse the University for the Permitted Costs;
- g. Subject to the foregoing and after consulting with the Associations, the University shall set the exact amount of the Fee after the conditions in section 3.01(a) and (b) are met, and before the Acceptance Date;
- h. If due to the timing associated with the initiation of the implementation of the Fee and delays in the Acceptance Date the University commences collecting the Fee for a period prior to the Acceptance Date, the Fee for such prior period shall be refunded to the Students from whom it was received; and
- i. Once established, the Fee shall not be indexed to inflation.

4.05 PAW Centre Account

The University shall establish and maintain as of the Acceptance Date a separate internal trust account designated as the PAW Centre Account. All funds received by the University from the imposition and collection of the Fees shall be deposited in the PAW Centre Account, and held in trust, so as to be administered and applied by the University for the purposes of and in accordance with the provisions of this Agreement. For greater clarity, notwithstanding the *Trustee Act*, RSA 2000 c. T-8, or any other applicable law related to trusts:

a. the University shall not have any obligations or be subject to any duties or restrictions with respect to the Fee and the PAW Centre Account, other than as are expressly set out in this Agreement;

- **b.** when the PAW Centre Account is in a credit balance, the University shall:
 - i. invest the balance in its non-endowed investment pool;
 - ii. invest the balance in accordance with the applicable investment guidelines which are established from time to time by the University for all non-endowment investments; and
 - be entitled to an administration fee, which shall be assessed and payable to the University as at the end of each month in which the PAW Centre Account is in a credit balance, as an allowance for its costs in relation to the PAW Centre Account, and shall be calculated by determining the month end balance in the PAW Centre Account and multiplying that balance by 0.042%.

The University shall on the last day of each month in which the PAW Centre Account is in a credit balance, report upon the investment income earned or losses incurred for the balance of funds within the PAW Centre Account for that month. Any such investment income or losses shall be credited or debited to and from a part of the balance of the PAW Centre Account.

4.06 Permitted Use of Student Fee Funding

The University shall debit or credit as appropriate against the PAW Centre Account, and to the extent funds are available within the PAW Centre Account, shall apply the same against the following (being collectively the "Permitted Costs"):

- **a.** Bridge Financing Costs not repaid from the proceeds of the Student Contributions Loan as contemplated in Section 4.02;
- **b.** all principal and interest payments pursuant to the Student Contributions Loan; and
- **c.** the following additional allowable costs in relation to the Student Contributions Loan:
 - i. prepayment charges, stop loss penalties, interest rate swap unwind costs, and all other costs fees, expenses, and charges payable by the University to the lender pursuant to the terms of the debenture, mortgage or interest rate swap agreement related to the Student Contributions Loan, where the incurring of the cost or penalty is shown to provide a cost saving to the Students as compared to the interest otherwise payable through the end of the then current fixed term of the Student Contributions Loan;
 - ii. carrying costs, provided that such costs are not duplicative to the interest and the principal of the Student Contributions Loan;

all payments and receipts pursuant to any interest rate swap agreement associated with the Student Contributions Loan, provided that the incurring of the cost is reasonable for the purpose of obtaining certainty as to the interest rate that will be obtained, as compared to the potential interest rate risk that would otherwise exist in association to the Student Contribution Loan; and provided further that the effective rate of interest or cost of borrowing calculated by combining such costs and the actual interest paid under the Student Contributions Loan over the term of the loan or the repayment, as the case may be, shall not in any event exceed the maximum rate of interest contemplated within section 3.01(a); and

d. the administration fee contemplated within Section 4.05(b)(iii).

4.07 University Responsible for Excess Costs

Notwithstanding any of the foregoing, the University shall be responsible for:

- a. the portion, if any, of Allowable Student Costs that exceeds the maximum principal amount of the Student Contributions Loan permitted under this Agreement; and
- **b.** all costs arising out of the Work which are not Allowable Student Costs, excluding any costs for which either or both the Students' Union and the GSA are responsible in accordance with the terms of this Agreement or the Lease Agreement.

4.08 No Rescission or Amendment

The payment or amount of the Fee may not be rescinded or amended by future referendums of the Students' Union or the GSA.

4.09 PAW Centre Account Accounting

The University shall establish and maintain separate internal accounting records with respect to the PAW Centre Account, the collection and disbursement of the Fee, and shall make such records available, on reasonable notice, for audit or examination by the Associations, their respective accountants or auditors, or other authorized representatives of the Students' Union or the GSA. The University shall maintain and dispose of such records in accordance with the relevant records retention schedule and disposal authority.

4.10 Annual Reporting

From and after the Acceptance Date, the University shall provide an annual report to the Strategic Operating Committee with respect to the collection and disbursement of the Fee, and the administration of all debits and credits from and to the PAW Centre Account.

PAW AD HOC STEERING COMMITTEE

5.01 Establishment

The Parties hereby establish the Steering Committee to serve as an ad hoc steering committee for duration of the conduct of the Work, and to work collaboratively and in good faith to advise the University on the development and promotion of policies, programming and strategic direction in relation to the Work as it relates to the Student Focused Areas

5.02 University-Committee Communications

The University shall advise the Steering Committee as to the progress of the Work in relation to the Student Focused Areas. If changes to the Work are required during this process, and such changes may affect matters within the Steering Committee's Scope, then the Steering Committee shall be advised and provide input with respect to the same.

5.03 Committee Scope and Procedures

The Parties agree that:

- a. the calling of meetings, the production of agendas, and the conduct of all business and affairs of the Steering Committee shall be subject to and in accordance with the Committee Procedures;
- b. the Steering Committee shall focus on advising the University on strategic matters, and shall not be involved in advising on the day-to-day details of the Work; and
- c. the University will at all times retain ultimate responsibility for the Work.

5.04 Committee Membership

The members of the Steering Committee shall be as follows:

- **a.** Vice-Provost and Dean of Students:
- **b.** Dean of the Faculty of Physical Education and Recreation;
- **c.** Associate Vice-President, Facilities and Operations;
- **d.** PAW Project Manager appointed by the Vice-President of Facilities and Operations;
- **e.** Representative appointed by the University Architect;
- **f.** Faculty of Physical Education and Recreation Project Coordinator;
- **g.** General Manager of the Students' Union;
- **h.** President of the Students' Union;
- i. Director of Finance and Operations of the GSA; and
- **i.** President of the GSA.

5.05 Member Proxies or Designates

Subject always to the Committee Procedures, each of the Steering Committee members listed above may appoint a proxy or designate to attend and participate in one or more meetings on behalf of such member, by providing written notice prior to the meeting or meetings to the Chair of the Steering Committee.

5.06 Committee Chair

The Chair of the Steering Committee shall be the Vice-Provost and Dean of Students.

5.07 Meetings

Meetings of the Steering Committee shall be held from time to time at such date and time and at such place as the Chair determines in accordance with Committee Procedures, and the conduct of such meetings shall be carried out by the Chair in accordance with Committee Procedures.

5.08 Quorum

A quorum for the transaction of business at any meeting of the Steering Committee shall consist of a majority of its members, provided always that a representative of each of the Students' Union and the GSA, or their respective designates or proxies, and one of the Dean of the Faculty of Physical Education and Recreation (or designate) or the Associate Vice-President, Facilities and Operations (or designate) are also in attendance. Notwithstanding any vacancy among the members of the Steering Committee, a quorum may exercise all the powers of the Steering Committee

5.09 Conflict

For the purposes of this section 5.09, a "conflict" means:

- a. any disagreement designated as a conflict by the University or by one of the Associations as to the conduct of the Steering Committee in accordance with the Steering Committee's Scope and this Agreement;
- **b.** any failure or alleged failure to comply with the Committee Procedures designated as a conflict by the University or by one of the Associations; or
- c. any matter, issue, decision or thing conducted or contemplated by the Steering Committee which is not agreed to by either of the Associations or the University and is so designated as a conflict by such Party.

Conflicts must be declared at the meeting of the Steering Committee at which the conflict occurs, or by written notice within Five (5) Business Days following the date of such meeting. In the event that a conflict occurs that cannot be resolved at the level of the Steering Committee, the

Party that designated such matter as a conflict at a meeting of the Steering Committee in accordance with the foregoing portion of this section 5.09 may refer the conflict for further discussion, review and resolution in accordance with the Conflict and Dispute Resolution Procedure.

5.10 Non-Member Attendees

Any other person may be admitted to a meeting of the Steering Committee on the invitation of the Chair or with the consent of the Steering Committee members at the meeting, for the purpose of providing input to assist the Steering Committee in advising on matters within the Steering Committee's Scope.

5.11 Dissolution

The Steering Committee shall be dissolved effective as of the Acceptance Date.

STRATEGIC OPERATING COMMITTEE

6.01 Establishment

The Parties hereby establish a Strategic Operating Committee, to work collaboratively and in good faith to advise and provide strategic guidance to the University for programming and activities in the Student Focused Areas, other than the Association Leased Areas, in keeping with the Shared Mission, the Project Purpose and Agreement Principles and in accordance with the Operational Requirements. Specifically, the Strategic Operating Committee shall:

- a. advise and provide input on operational policies for consideration by the University and the Associations so as to provide for efficient and effective interactions between the Parties as occupants of the Project, and the efficient and effective use of space available from time to time throughout the Project;
- b. receive regular reports from University staff who manage the PAW Centre and any Other Renovated Areas in which student programming is taking place regarding programming, activities, and general operating budgets and costs within the Student Focused Areas, other than the Association Leased Areas;
- c. recommend program principles to be used by the University in the development of the annual operating budget related to the Student Focused Areas, other than the Association Leased Areas;
- **d.** periodically review, and make recommendations about, the long-term strategic direction of programming and activities within the Student Focused Areas, other than the Association Leased Areas;
- e. in making its recommendations, and in a manner that is consistent with the Shared Mission, the Agreement Principles and the Project Purpose, endeavour to

maximize revenue-generating opportunities in relation to the PAW Centre and Other Renovated Areas, so that such funds can be used by the University to cover its costs in relation to operating the PAW Centre and Other Renovated Areas and enhance programming and operations of the PAW Centre and Other Renovated Areas; and

f. review all collection and disbursement information about the Fee, and the PAW Centre Account.

6.02 Commencement

The Strategic Operating Committee shall begin meeting on or after the Acceptance Date.

6.03 Committee Scope and Procedures

The Parties agree that:

- a. the calling of meetings, the production of agendas, and the conduct of all business and affairs of the Strategic Operating Committee shall be subject to and in accordance with the Committee Procedures;
- **b.** the Strategic Operating Committee shall focus on advising the University on strategic matters, and shall not be involved in overseeing or advising on the day-to-day details of the management and operation of the PAW Centre or the Other Renovated Areas;
- c. the University will at all times retain ultimate responsibility for all Operation and Maintenance of the PAW Centre and the Other Renovated Areas, excluding the Association Leased Space.

While the role of the Strategic Operating Committee is advisory only and the University retains the right to determine its priorities and allocate its resources, the expectation of the Parties is that normally the University will not unreasonably reject advice, input, or recommendations of the Strategic Operating Committee that are supported by the members identified in sections 6.04 a., b., e., f., and h.

6.04 Membership

The members of the Strategic Operating Committee shall be as follows:

- **a.** The Dean of the Faculty of Physical Education and Recreation;
- **b.** The Vice-Provost and Dean of Students;
- **c.** The Facility Manager of the Fitness Centre;
- **d.** The Director of Operations for the Faculty of Physical Education and Recreation;
- **e.** A representative appointed by the Vice-President (Facilities and Operations);
- **f.** The President of the GSA, or his or her designate;

- **g.** One (1) representative or appointee appointed by the Board of the GSA; and
- **h.** The President of the Students' Union, or his or her designate;
- **i.** Four (4) representatives or appointees appointed by the Students' Council of the Students' Union.

6.05 Member Proxies or Designates

Each of the Strategic Operating Committee members listed above may appoint a proxy or designate to attend and participate in one or more meetings on behalf of such member, by providing written notice prior to the meeting or meetings to the Chair of the Strategic Operating Committee.

6.06 Chair

The Chair of the Strategic Operating Committee shall be the Vice-Provost and Dean of Students.

6.07 New Members

New members may be added to the Strategic Operating Committee from time to time at the direction of the Strategic Operating Committee provided always that the number of members appointed by the Associations shall always exceed the number of other members of the Strategic Operating Committee.

6.08 Meetings

Meetings of the Strategic Operating Committee shall be held a minimum of three times per year, at such date and time and at such place as the Chair determines in accordance with Committee Procedures, and the conduct of such meetings shall be carried out by the Chair in accordance with Committee Procedures.

6.09 Quorum

A quorum for the transaction of business at any meeting of the Strategic Operating Committee shall consist of a majority of its members, provided always that:

- **a.** the President of the GSA, or his or her proxy or designate, is in attendance;
- **b.** the President of the Students' Council, or his or her proxy or designate, is in attendance; and
- c. either the Dean of the Faculty of Physical Education and Recreation (or designate), or the representative appointed by the Vice-President (Facilities and Operation) (or designate) is also in attendance.

Notwithstanding any vacancy among the members of the Strategic Operating Committee, subject to the foregoing a quorum may exercise all the powers of the Strategic Operating Committee.

6.10 Conflict

For the purposes of this section 6.10, a "conflict" means:

- a. any disagreement designated as a conflict by the University or by one of the Associations as to the conduct of the Strategic Operating Committee in accordance with the Strategic Operating Committee's Scope and this Agreement;
- **b.** any failure or alleged failure to comply with the Committee Procedures designated as a conflict by the University or by one of the Associations; or
- any matter, issue, decision or thing conducted or contemplated by the Strategic Operating Committee which is not agreed to by either of the Associations or the University and is so designated as a conflict by such Party.

Conflicts must be declared at the meeting of the Strategic Operating Committee at which the conflict occurs, or by written notice within Twenty (20) Business Days following the date of such meeting. In the event that a conflict occurs that cannot be resolved at the level of the Strategic Operating Committee, the Party that designated such matter as a conflict in accordance with the foregoing portion of this section 6.10 may refer the conflict for further discussion, review and resolution in accordance with the Conflict and Dispute Resolution Procedure.

6.11 Non-Member Attendees

Any other person may be admitted to a meeting of the Strategic Operating Committee on the invitation of the Chair or with the consent of the Strategic Operating Committee members at the meeting, for the purpose of providing input to assist the Strategic Operating Committee in advising and providing strategic guidance on matters within the Strategic Operating Committee's Scope. Notwithstanding the foregoing, the Students' Union and the GSA are each entitled to permit the attendance of a non-voting attendee, designated or invited by each of the respective Associations from time to time, at all meetings of the Strategic Operating Committee.

OPERATION, MANAGEMENT, AND ACCESS

7.01 Operation and Maintenance

The University shall operate, manage and maintain the PAW Centre and Other Renovated Areas, other than Association Leased Space, during the Term in accordance with the terms of this Agreement including the Operational Requirements.

7.02 University Policies

The use of the PAW Centre and Other Renovated Areas shall be subject to:

- a. such policies and procedures of general application to use of buildings owned by the University as the University deems necessary, acting reasonably, in order to ensure the safety and security of persons and property, to give effect to the *Code of Student Behaviour* or other policies or procedures relating to student discipline, to comply with applicable legislation, or for other purposes of similar importance;
- b. such other policies and procedures of general application to use of buildings owned by the University as the University deems appropriate, provided that such policies or procedures will not in any event contravene, or cause or contribute to a contravention of any of the terms of this Agreement including, without restriction, the Shared Mission, the Agreement Principles, the Project Purpose, and the Operational Requirements; and
- c. such policies and procedures, rules and regulation governing the use of the PAW Centre and Other Renovated Areas specifically as the University deems appropriate, taking into consideration the advice and strategic guidance of the Strategic Operating Committee, provided always that such policies, procedures, rules or regulations, and any amendments thereto which apply only to the PAW Centre and Other Renovated Areas and are material in nature:
 - i. will be submitted to the Strategic Operating Committee at a regularly scheduled meeting prior to implementation unless the University acting reasonably determines that implementation is required prior to submission in order to ensure safety and security of persons and property in which case the same will be submitted to the Strategic Operating Committee as soon as reasonably possible following implementation;
 - ii. will not in any event contravene, or cause or contribute to a contravention of any of the terms of this Agreement including, without restriction, the Shared Mission, the Agreement Principles, the Project Purpose, and the Operational Requirements.

7.03 Student Use Entitlements

Subject to cessation of the Fee in accordance with Section 11.02 of this Agreement, Students who have paid the Fee for a Fall Term, Spring Term, Summer Term or Winter Term are entitled to use the Student Focused Areas during the regular hours of operation throughout the respective Fall Term, Spring Term, Summer Term or Winter Term, as the case may be, free of any other charge, subject to:

a. the terms and conditions of this Agreement and the Lease Agreement;

- **b.** the University's ability to charge fees to Students and others in association with locker rentals;
- c. the use, programs, rentals, and maintenance described in sections 7.04 and 7.05 hereto, which will have priority over the relevant portion of the Student Focused Areas during the time period that such programs, rentals or maintenance are taking place.

No provision of this Agreement permits a person who has been excluded from access to the PAW Centre and Other Renovated Areas whether alone or in conjunction with other facilities of the University and whether pursuant to policies and procedures of the University or by other lawful action to use the Student Focused Areas.

7.04 Booking Process

The University shall in its management of the PAW Centre and other Renovated Areas establish and operate a reasonable and fair space booking system (the "Booking System"). In addition to any other stated subsidiary goals the primary goal of the Booking System is to provide the University with the ability to permit utilization of the Bookable Facilities by the University and a broad array of students, student groups and others in the communities it serves during those periods in which the Bookable Facilities are not otherwise booked for use through priority booking rights of the Associations thereby maximizing the use of the Bookable Facilities. Unless otherwise agreed between the Parties, neither of the Associations nor a student group registered with the University will pay a rental fee for use of the Bookable Facilities, but may be required to reimburse the University for costs incurred by the University with respect to cleaning a Bookable Facility, or other costs incurred as a result of such Association or student group's use of such Bookable Facility. The Booking System shall:

- a. include reasonable provisions permitting the Associations to exercise priority rights pursuant to the Booking System both to the long-term bookings and the short-term bookings of Bookable Facilities;
- b. provide for at least an annual meeting, or if requested by the Associations one meeting per Term, between the appropriate representatives of the University and of each of the Associations at which firstly the Associations may exercise their long-term priority booking rights for the forthcoming period and secondly the University and the Associations may review the process for the Associations to exercise their short-term priority booking rights for the forthcoming period;
- **c.** provide for a balance between long-term booking and bookings on a short-term basis; and
- d. recognize that although the Graduate Student Lounge/Quiet Study is included within Bookable Facilities the intended primary use of that space is as quiet study space and accordingly should be capable of being booked through the Booking System only on an occasional basis.

The revenue expected to be generated from use of Bookable Facilities while desirable is not expected to be significant in terms of off-setting operating costs of the Bookable Facilities and is not defined as either the primary or a subsidiary goal of the Booking System.

7.05 Programs

Subject always to the Operational Requirements, the University may:

- a. provide campus recreation or other instructional fitness classes in the Fitness Centre Area, the Climbing Centre Area or, subject to Section 7.04, in the Student Focused Areas other than Association Leased Space, and students who wish to participate in such classes shall pay the required fees for such class, separate and apart from the Fee, notwithstanding section 7.03;
- b. subject to section 7.04 herein, enter into a licence agreement with a third party to allow it to access space within the Student Focused Area, other than the Association Leased Space, for its own use for a certain period of time; and
- c. perform maintenance as may be required from time to time within the Student Focused Areas, as set out in section 9.01.

7.06 Academic Staff and Non-Academic Staff Users

The University may permit Academic Staff and Non-Academic Staff contributing to the operational costs of the Project through their various association fees access rights similar to Students' access as described in section 7.03 during regular hours of operation, and subject to all additional fees as outlined in 7.05.

7.07 Other Users

Subject always to the Operational Requirements, the University may permit other individuals who are not Students, Academic Staff or Non-Academic Staff, to use the Student Focused Areas and may in its discretion charge such individuals a usage fee, in order to pay for operating, programming, capital renewal, building reserves, and other costs associated with the PAW Centre and Other Renovated Areas incurred within the Term.

7.08 Minimum Hours

Minimum hours of operation of and access to the Student Focused Areas other than Association Leased Space shall be in accordance with University policy, subject always to the requirements of the Operational Requirements.

ASSOCIATION LEASED SPACE

8.01 Lease

The Parties shall execute and deliver a lease for the space designated as "Food Services Vendor" in the Design Development Report, which shall form a part of the Association Leased Space on the terms and conditions set out in the Lease Agreement concurrently with the execution and delivery of this Agreement. At the written request of the Associations to the University made no earlier than the Acceptance Date and no later than one year following the Acceptance Date the Parties shall execute and deliver a lease for the Pro Shop Space in the Design Development Report on the terms and conditions set out in the Lease Agreement, amended as follows:

a. The following sentence shall be added at the end of paragraph 4.01 of such Lease Agreement:

"Notwithstanding the foregoing, the Tenant shall consult with the Faculty of Physical Education and Recreation prior to using the Premises for any purpose other than the operation of a Pro Shop."

MAINTENANCE AND REPURPOSING

9.01 Operation

During the Term, the University shall maintain the PAW Centre and the Other Renovated Areas, including without limitation the Common Areas, and the improvements, equipment, furniture, fixtures, University signs, and other incidentals contained in the PAW Centre and Other Renovated Areas, other than the Association Leased Space and other than any improvements, equipment, furniture, fixtures, signs, and other incidentals located in the Association Leased Space or located elsewhere in the PAW Centre and Other Renovated Areas installed by or on behalf of the Associations, contained in the PAW Centre and Other Renovated Areas, in accordance with the Operational Requirements.

9.02 Change of Use

During the Term:

- a. the Associations may make a request to the University to change the use for a particular area within the Student Focused Areas, other than Association Leased Areas. If such a change is approved by the University and if Alterations to such an area would be required in order to achieve such a change, then the University:
 - i. shall perform or arrange to be performed such Alterations;
 - ii. shall purchase and install any new improvements, equipment, furniture, fixtures, signs and other incidentals required in order to achieve such a change; and
 - iii. may levy a charge for overheads associated with services performed by the University;

- at the sole cost and expense of Students' Union and the GSA and all such costs and expenses shall be paid by the Students' Union and the GSA, within 30 days of receiving an invoice for the same from the University;
- b. Notwithstanding section 9.02 (a), the University may, at its sole discretion, agree to join with the Students' Union and the GSA in a project to achieve such a change and may pay for a portion of the costs of such a change.

9.03 Repurposing of Student Focused Areas

Student Focused Areas, or any portion thereof, shall not during the initial thirty-five (35) years of the Term be converted to academic, instructional, research, or administrative uses, nor any other use not contemplated within this Agreement, without the prior consent of the Associations.

PAW CENTRE ACCOUNT SURPLUS

10.01 Use of Surplus

Any funds remaining in the PAW Centre Account following the full repayment of Permitted Costs shall, except as otherwise provided in section 13.04, be allocated to capital improvements or enhancements to:

- **a.** equipment and features contained within the Fitness Centre Area of the PAW Centre; or
- **b.** the Student Focused Areas:

in a manner agreed to by the Parties.

TERM OF AGREEMENT

11.01 Term

This Agreement shall become of force and effect from the date of execution on behalf of all Parties, and unless sooner terminated by mutual consent of the Parties or pursuant to section 12.04 hereof shall continue in effect for an aggregate term of:

- a. thirty-five years commencing on Acceptance Date; and
- b. the period thereafter that ends on the date so specified in a notice from the University to the Associations, which date shall be not less than six (6) months following the giving of such notice, which notice states that the University has, in accordance with its then applicable procedures and processes determined that:

- i. a substantial portion of the Student Focused Areas cannot continue to be used for the purposes for which they were intended in the absence of substantial renovation or upgrading; or
- ii. a substantial portion of the Student Focused Areas should be repurposed to other uses not contemplated within this Agreement.

Notwithstanding the foregoing portion of this section 11.01:

- c. if the University makes the determination referred to in section 11.01(b) it shall prior to giving notice of termination pursuant to this section 11.01 as a result of such determination advise the Associations of its determination and engage in meaningful consultations with the Associations which shall include an opportunity for the Associations to make such proposals as the Associations deem appropriate for the funding of such renovations or upgrading by the Associations;
- **d.** prior to, or as a part of, making the determination referred to in section 11(b) the University shall engage in meaningful consultations with the Associations with respect to such matter;
- e. any of the Parties may as a part of the consultations on a matter contemplated in this section 11.01 require that the matter be referred to the Strategic Operating Committee to permit it to provide its advice to the University;
- in this section 11.01 a substantial portion of the Student Focused Areas includes the Fitness Centre Area and the Climbing Centre Area.

11.02 Repayment of Permitted Costs

Upon repayment in full of the Permitted Costs:

- a. all other terms of this Agreement unrelated to such repayment shall continue for the balance of the Term;
- b. without restricting the foregoing, the Students shall continue for any balance of the Term to have access to the Student Focused Areas other than the Association Leased Space in the manner set out in this Agreement; and
- **c.** the payment of the Fee shall cease.

For greater certainty, nothing in this Agreement prevents the University from continuing to charge to students the athletics and recreation fee or its equivalent, in accordance with University policy, as amended from time to time.

11.03 Termination

After this Agreement is terminated, all space in the PAW Centre and Other Renovated Areas shall be governed and managed pursuant to applicable University policies and procedures, as amended, replaced or supplemented from time to time.

DAMAGE AND DESTRUCTION REPAIR AND RESTORATION

12.01 Damage or Destruction

In the event of:

- **a.** the substantial damage rendering 40% or more of the PAW Centre and Other Renovated Areas unusable; or
- **b.** the destruction of the PAW Centre or Other Renovated Areas;

by fire or by any other casualty requiring more than 180 days to repair or reconstruct, the Steering Committee shall be revived, and it shall be governed by sections 5.03(a) and 5.04 through 5.10 inclusive of this Agreement. The Steering Committee shall meet to review the current business model, space program, and outstanding obligations in association with the Student Contribution Loan, and shall consider whether the business model and space program can be modified in such a way that it will be financially viable to repair and rebuild the damaged portions of the PAW Centre and Other Renovated Areas.

12.02 Committee Review

In the event of the substantial damage to or destruction of the PAW Centre and Other Renovated Areas as defined above, and in the event that the Steering Committee determines that the business model and space program can be modified in such a way that it will be financially viable to repair and rebuild the damaged portions of the PAW Centre and Other Renovated Areas, then the Steering Committee shall make a recommendation to the University about how to repair and rebuild the PAW Centre and Other Renovated Areas within 180 days of the occurrence of the damage or destruction.

12.03 Reconstruction

In the event of:

- **a.** the damage to the PAW Centre and Other Renovated Areas is less than substantial as defined above; or
- b. substantial damage to or destruction of the PAW Centre and Other Renovated Areas as defined above, and the University determines that it is financially viable to repair and rebuild the PAW Centre and Other Renovated Areas;

the University shall repair and rebuild the same, provided that any preconditions for financial viability it has determined are necessary have taken place. The University's obligation to repair

and rebuild shall not include the obligation to repair or rebuild any chattel, fixture, leasehold improvement, installation, addition or partition in respect of which the Students' Union, the GSA, or a sub-lessee of the Students' Union and/or the GSA is required to maintain insurance.

12.04 No Reconstruction

In the event of the substantial damage to or destruction of the PAW Centre and Other Renovated Areas as defined above and the University determines that it is not financially viable to repair and rebuild the PAW Centre and Other Renovated Areas, then the University shall cease to collect the Fee and subject to the provisions of section 13.04 this Agreement shall cease.

12.05 Reconstruction by Associations

If a chattel, fixture, leasehold improvement, installation, addition or partition in respect of which the Students' Union and the GSA are required to maintain insurance is damaged or destroyed by fire or by any other casualty, and if the PAW Centre and Other Renovated Areas is rebuilt in accordance with section 12.03, then the Students' Union and the GSA shall repair, rebuild or replace the same.

INSURANCE

University Insurance

The University shall keep in full force and effect, at its expense:

- a. commercial general liability insurance in an amount of not less than five million dollars (\$5,000,000) in respect of injury to or death of one or more persons or property damage against claims for personal injury liability, death or property damage occurring on, in or about the PAW Centre or Other Renovated Areas, including personal liability, (which policy shall provide for cross-liability and severability of interests);
- b. insurance against loss or damage by fire or other perils to the PAW Centre building and Other Renovated Areas and equipment appurtenant thereto to the full replacement value of same and containing the standard extended perils and endorsements, PROVIDED HOWEVER, that this covenant shall not impose any obligation upon the University to insure any of the Students' Union, GSA or any Sub-Tenant's fixtures, merchandise, stock-in-trade, furniture or other improvements; and
- **c.** boiler and machinery insurance with respect to all boilers and machinery owned or operated by the University.

The foregoing policies shall be kept in good standing and in full force and effect at all times throughout the Term.

13.01 Association Insurance

The Students' Union and GSA shall each maintain at its sole cost insurance as follows:

- a. general public liability insurance against claims for personal injury, death or property damage arising out of the operations of the named insured with respect to its obligations assumed under this Agreement, such insurance to afford protection to the combined single limit for personal injury, death or property damage of not less than two million (\$2,000,000.00) Dollars or such other limits as may be approved or requested by the University;
- **b.** other insurance as set out in the Lease Agreement.

The foregoing policies shall be kept in good standing and in full force and effect at all times throughout the Term.

Use of Proceeds

Subject to section 12.04, the approval of any lender to whom an outstanding debt is owed in relation to the construction of the PAW Centre or Other Renovated Areas, and the balance of this section, the proceeds of all property coverage insurance received by the University in relation to the damage or destruction of the PAW Centre or Other Renovated Areas shall be utilized for the repair and reconstruction of the PAW Centre or Other Renovated Areas. In the event of the decision of the University to not repair or rebuild the PAW Centre and Other Renovated Areas as contemplated under section 12.04 hereto, the University shall apply the proceeds of insurance received by the University from the insurance maintained by the University pursuant to section 13.01(b) with regards to the PAW Centre and Other Renovated Areas as follows:

- **a.** Firstly, for the repayment of the Student Contributions Loan and any other outstanding Permitted Costs;
- **b.** Secondly, for the payment of all direct demolition and site remediation costs incurred by the University in demolishing the damaged or destroyed portions and restoring the applicable lands to a safe condition and ready for development for alternative uses;
- c. Thirdly, deposit in the PAW Centre Account a percentage of the balance equal to the portion that the capital contribution to the PAW Centre and the Other Renovated Areas made by the students through repayment of capital of the Student Contributions Loan through the PAW Centre Account is of the total capital costs of the PAW Centre and/or the Other Renovated Areas, as the case may be;
- **d.** Lastly, distribute the balance to the University.

PAW Centre Fund Surplus

If, following application of insurance proceeds as contemplated in section 13.03, there is a balance of funds remaining in the PAW Centre Account, any funds remaining in the PAW Centre Account shall be used by the University for such non-curricular, health, fitness and wellness-related student activities as agreed upon between the University and Associations, each acting reasonably. If the University and the Associations are unable to agree upon such non-curricular, wellness-related student activities within eighteen (18) months following the deposit of funds pursuant to section 13.03 in the PAW Centre Account either the University or the Associations may require that the failure to agree be treated as a dispute which will be resolved as contemplated in **Schedule F-1 (Conflict and Dispute Resolution Procedure)** subject to the following:

- **a.** If the dispute is not resolved through the Negotiation or Mediation it shall be determined through arbitration;
- b. The Associations shall submit their common proposed non-curricular, wellness-related student activities, and the University shall submit its proposed non-curricular, wellness-related student activities, for which the funds are to be used to the arbitrator who shall choose between such submissions;
- **c.** The arbitrator shall be guided by, and the decision shall be respectful of, the Shared Mission, Agreement Principles and Project Purpose; and
- **d.** Neither the University nor the Associations shall submit proposed non-curricular, health, fitness and wellness-related student activities the implementation of which would impose costs or expenses on the other.

LIMITATION OF LIABILITY AND INDEMNITY

14.01 Limitation of Liability

Notwithstanding anything to the contrary expressed or implied in this Agreement, each Party and its respective directors, officers, employees or agents shall not be liable to any other Party for any general, indirect, punitive, exemplary, or consequential damages or any economic losses of any kind, including without limitation loss of profits, loss of opportunity, loss of production, or loss of use, regardless of whether the liability to which such damages relate arises in contract, tort or otherwise in law.

14.02 Indemnity

Except as otherwise specified in section 14.01, each Party (the "Indemnifier"):

- a. shall be liable to each of the other Parties (including their respective directors, officers, employees and agents), (the "Indemnified Party"); and
- **b.** shall indemnify and hold harmless the Indemnified Party from and against:

any and all liabilities, claims (including third party claims), demands, fines, charges, suits or actions, and without limiting the generality of the foregoing, any direct losses, costs, damages and expenses of the Indemnified Party, including costs as between a solicitor and his own client (all together referred to as "Damages") which may be brought or made against the Indemnified Party or which the Indemnified Party may pay, suffer or incur as a result of, or arising out of:

- **c.** any breach or default of this Agreement by the Indemnifier,
- d. the negligence or wilful misconduct of the Indemnifier in the performance of its obligations under this Agreement,

to the extent that the Damages are caused by the breach, default, negligence or wilful misconduct of the Indemnifier. These indemnities shall survive the expiration or termination of this Agreement.

DEFAULT AND REMEDIES

15.01 Notification

If either the Associations or the University does not:

- a. within a period of twenty (20) Business Days of notice of demand from the other remedy any failure on its part to perform or observe any material term or condition of this Agreement; or
- where more than twenty (20) Business Days is reasonably required to remedy such failure, within a period of twenty (20) Business Days commence and thereafter diligently pursue the remedy of such failure;

such of the Associations or the University that demanded the other remedy such failure may require the matter be treated as a dispute which will be resolved as contemplated in **Schedule F-1 (Conflict and Dispute Resolution Procedure)** subject to the following:

- c. if the Dispute it not resolved through the Negotiation or Mediation it shall be determined through arbitration;
- d. in addition to any other rights, remedies, directions or orders the arbitrator is entitled to give or impose the arbitrator may in the case of a failure on the part of the University to perform or observe any material term or condition of the Agreement order that any Fees subsequently collected by the University from and after such order shall be held in a separate account and not deposited to the PAW Centre Account until the applicable default has been remedied and the application of any balance of funds remaining in the PAW Centre Account to Permitted Costs be suspended until the applicable default has been remedied:

e. any interim costs incurred due to the unavailability of the application of funds from the PAW Center Account to Permitted Costs as contemplated within subsection d. above shall be the sole responsibility of the University.

CONFLICT AND DISPUTE RESOLUTION

16.01 Conflicts and Disputes

In the event that a conflict arises as contemplated within section 6 of Schedule B-1 or section 3.04, 5.09 or section 6.10 of this Agreement, (referred to in this section 16.01 and **Schedule F-1** (Conflict and Dispute Resolution Procedure) as a "Conflict"), or a dispute, controversy or claim otherwise arises out of, in connection with, or relating to this Agreement, or the breach, termination or invalidity thereof (referred to in this section 16.01 and **Schedule F-1** (Conflict and Dispute Resolution Procedure) as a "Dispute"), the Conflict or Dispute, as the case may be, shall be resolved between the Parties in accordance with the Conflict and Dispute Resolution Procedure.

FOIP AND CONFIDENTIALITY

17.01 Confidentiality

The Students' Union and the GSA acknowledge that the University is a public body subject to the *Freedom Of Information and Protection of Privacy Act* of Alberta ("FOIPP Act"), as amended from time to time, or any successor legislation then in force, and that access to information in respect of this agreement will be provided in accordance with the FOIPP Act. The records transferred to or collected, created, maintained or stored under this agreement for the University are subject to the access provisions of the FOIPP Act. If the University receives a request for any of the records that are in the custody of the Students' Union or the GSA, it will be the responsibility of such Party to provide copies of the records, at such Party's expense, subject always to the processes, exemption, and rights to object contemplated within the FOIPP Act. The Students' Union or the GSA, as the case may be, must provide such copies to the University's Information and Privacy Office within 10 calendar days from notification by the University's Information and Privacy Office.

17.02 Disclosure

Subject to section 17.03, each of the University, the Students' Union and the GSA may disclose information it considers confidential to the other pursuant to this Agreement and, without limitation, the committees established pursuant to this Agreement. Each Party will use all reasonable efforts to treat and keep confidential, and cause its executives, officers, appointees and employees to treat and keep confidential, any such information received by it from another Party marked or otherwise identified as confidential. Such reasonable efforts will be no less than the efforts used by the recipient to protect its own confidential information.

17.03 Exceptions

The obligation to keep information confidential will not apply to information:

- **a.** which was already known to the Party to whom it was disclosed;
- **b.** when such information is received by the recipient from a third party without a duty of confidentiality;
- c. when the Party is ordered by judicial or administrative process to disclose any or all of the information, in connection with legal proceedings relating to this Agreement, or when a party is required by law to disclose such information; or
- **d.** when the other party gives prior written consent to disclosure,

and will cease to apply when, through no fault or omission of the receiving party or its officers or employees, such confidential information becomes public knowledge.

GENERAL

18.01 Notices

Subject to Article 11 of Schedule E-1, any notice required or permitted to be given to any party hereunder shall be validly given if either delivered to the address provided below (if not a post office box), or by electronic facsimile to the fax number provided below:

The University

Insert Address Fax: Insert

Students' Union

Insert Address Fax: Insert

GSA

Insert Address Fax: Insert

or to such other address as a party may from time to time direct in writing. Any such notice delivered or faxed by confirmed fax as aforesaid shall be deemed to have been received by and given to the addressee when delivered or faxed.

18.02 No Assignment

No party shall assign or transfer this Agreement or any of their rights or obligations arising hereunder without the prior written consent of the other Parties.

IN WITNESS WHEREOF the duly authorized officers of the Parties have executed and delivered this Agreement on the dates indicated below.

THE GOVERNORS OF THE UNIVERSITY OF ALBERTA	
Per:	Date:
Vice-President (Finance and Administration)	
Per:	Date:
Vice-President (Facilities and Operations)	
Per:	Date:
Provost and Vice-President Academic	
THE STUDENTS' UNION OF THE UNIVERSITY OF ALBERTA	THE GRADUATE STUDENTS' ASSOCIATION OF THE UNIVERSITY OF
ALBERTA	
Per:	Per:
President	President
Per:	Per:

	PA	W	CENTRE	AGREEMEN	ľΠ
--	----	---	--------	-----------------	----

		_
Date	Date	

SCHEDULE A-1 – DEFINITIONS

In this Agreement, unless something in the subject matter or context is inconsistent therewith:

- **a.** "Academic and Research Areas" means the Steadward Centre Area, the Faculty of Physical Education and Recreation Area, the Renewed Areas, and all other areas of the PAW Centre and Other Renovated Areas used primarily by the University for teaching or research purposes;
- **b.** "Academic Staff" means faculty and other individuals directly employed with the University and constituting "Academic Staff" defined as within the *Post-secondary Learning Act*;
- **c.** "Acceptance Date" means the date determined in accordance with section 3.04 of this Agreement;
- **d.** "Agreement Principles" has the meaning set out in Recital D;
- e. "Allowable Student Costs" means:
 - i. the costs of planning, designing, constructing/renovating, and commissioning of the Student Focused Areas:
 - ii. the initial costs of purchasing and installing equipment, furniture, signs, fixtures, and other incidentals to be used in the Student Focused Areas; and
 - the percentage of the costs of planning, designing, constructing and commissioning the Common Areas as part of the Work that is equal to the percentage that is calculated as the total net assignable square metres occupied by the Student Focused Areas, divided by the total net assignable square metres in the Project that are occupied by the Student Focused Areas and the Academic and Research Areas, which method of calculation and allocation is to be consistently applied, so as to fairly and equitable allocate such costs across the constituent portions of the Project including, without restriction, the Student Focused Areas and the Academic and Research Areas:
- **f.** "Alterations" means all alterations, renovations, changes, improvements, reconfigurations, retrofitting, modifications, substitutions, additions, removals, expansions and reductions;
- **g.** "Associations" means, collectively, the Students' Union and the GSA;
- **h.** "Association Leased Space" means all that portion of the Student Focused Areas designated or otherwise described as follows within the Design Development Report;

- i. the Food Services Vendor space;
- ii. the Pro Shop Space;
- **i. "Board"** means the board of the GSA, as appointed or otherwise established from time to time;
- **j.** "Bookable Facilities" means those facilities designated or otherwise described as follows within the Design Development Report:
 - i. the Community/Teaching Kitchen;
 - ii. the Multi-Purpose Room; and
 - iii. the Graduate Student Lounge which is to be subsequently designated as the Graduate Student Lounge/Quiet Study;

together with such additional areas as the Parties may agree upon from time to time;

- **k.** "Bridge Financing Costs" has the meaning set out in section 4.01;
- **l.** "Climbing Centre Area" means all that portion of the Student Focused Areas designated or otherwise described as the Climbing Centre area within the Design Development Report;
- m. "Committee Procedures" means those procedures contained within Schedule E-1 (Committee Procedures), as amended from time to time as provided therein or otherwise by agreement in writing between the Parties;
- **n.** "Common Areas" means all areas within the PAW Centre that do not fall within the definitions of Academic and Research Areas and Student Focused Areas, including without limitation:
 - i. roofs, roof decks, roof coverings and roof membranes; exterior wall assemblies including without limitation weather walls, exterior and interior structural components and elements (including structural subfloors and structural ceilings) and beams, pillars and bearing and retaining walls;
 - ii. foundations and footings:
 - iii. common restroom and washroom facilities not located directly within Academic and Research Areas or Student Focused Areas;
 - iv. walkways, hallways, stairs, elevators, escalators, other transportation equipment and systems, delivery passages, doors, loading docks and related areas, communal garbage and recycling areas and facilities, pedestrian sidewalks, pedways, tunnels, corridors, passageways, accesses, terraces and walkways, service and fire corridors, lamp standards, and entrances to and exits from the PAW Centre, which are not for the exclusive benefit of users of an Academic and Research Area or a Student Area;

- **v.** base building services, including but not limited to mechanical, electrical, utilities, and equipment and service rooms;
- vi. fire prevention, fire detection, alarm and security systems and equipment; and
- vii. landscaping and site development adjacent to the PAW Centre;
- **o.** "Concourse Area" means all that portion of the Project designated or otherwise described as the Concourse area within the Design Development Report
- **p.** "Condition Day" has the meaning set out in section 3.01 hereto;
- **q.** "Design Development Report" means the University of Alberta Physical Activity & Wellness (PAW) Centre Design Development Report dated October 13, 2011, prepared by Group 2 Architecture Engineering Ltd.;
- **r.** "Faculty of Physical Education and Recreation Area" means all that portion of the Project designated or otherwise described as the Faculty of Education and Recreation space within the Design Development Report.
- **s. "Fall Term"** means each fall term as defined in the University's calendar, as amended from time to time;
- **t.** "Fee" means a non-instructional fee levied on each undergraduate and graduate Student of the University, subject to the limitations and terms set out in this Agreement;
- **u.** "Fitness Centre Area" means all that portion of the Student Focused Areas designated or otherwise described as the Fitness Centre area within the Design Development Report;
- v. "Functional Requirements" means the functional requirements and specifications set forth within Schedule B-1(Functional Requirements);
- w. "GSA" means the Graduate Students' Association of the University of Alberta;
- **x.** "GSA Council" means council of the GSA elected from time to time by its membership;
- y. "Lease Agreement" means a lease agreement(s), substantially in the form attached hereto as Schedule D-1 (Form of Lease Agreement);
- **z.** "Non-Academic Staff" means individuals directly employed with the University and constituting "Non-Academic Staff" defined as within the *Post-secondary Learning Act*;
- aa. "Operational Requirements" means the provisions set out in Schedule C-1 (Operational Requirements);

- **bb.** "Operation and Maintenance" means management, operation and maintenance of the PAW Centre and the Other Renovated Areas, including:
 - i. provision of all utilities and services;
 - ii. provision of heating and lighting;
 - **iii.** provision of service staff, security staff, janitorial staff, and building operations and maintenance staff;

in accordance with this Agreement;

- **cc.** "Other Renovated Areas" means areas within the existing Van Vliet Physical Education and Recreation Centre at the University comprising Student Focused Areas, as modified from time to time in accordance with the provisions of this Agreement;
- **dd.** "Party" means one of the University, the GSA or the Students' Union, and "Parties" means all of them;
- **ee. "PAW Centre"** means the new building to be constructed in accordance with the Work, as modified from time to time in accordance with the provisions of this Agreement;
- **ff. "PAW Centre Account"** means the account established and maintained by the University as provided in section 4.05;
- **gg.** "Permitted Costs" has the meaning set out in s. 4.06;
- **hh.** "**Project**" means the planning, design, construction/renovation and commissioning of:
 - i. the PAW Centre and Other Renovated Areas; and
 - ii. any other physical structures and improvements the construction or renovation of which are contemplated in the Design Development Report and which are constructed/renovated by the University in conjunction with the construction or renovation of the PAW Centre and Other Renovated Areas, those contemplated structure and improvements including:
 - A. the Faculty of Physical Education and Recreation Area; and
 - **B.** the Steadward Centre Area.
- ii. "Project Purpose" has the meaning set out in section 1.05;

- **jj.** "**Pro Shop Space**" means the area designated as the Activity Registration Zone, Student/Faculty Pro Shop area shown within the Design Development Report, excluding an area on the northern section of the Activity Registration Zone, Student/Faculty Pro Shop, which is:
 - i. located adjacent to the lower concourse;
 - ii. approximately 15 square metres, more or less, with the exact perimeters to be determined by the University;
 - iii. to be used by the University as the Activity Registration Zone, as such use may be amended from time to time in the University's discretion;

as such area may be amended pursuant to the terms of the applicable Lease;

- **kk.** "Renewed Areas" means all that portion of the Project designated or otherwise described as the Renewed Space and the Renewed Additional Space within the Design Development Report;
- **Il.** "Shared Mission" has the meaning set out in Recital A;
- **mm.** "Spring Term" means each spring term as defined in the University's calendar, as amended from time to time:
- **nn.** "Steadward Centre Area" means all that portion of the Project designated or otherwise described as the Steadward Centre area within the Design Development Report;
- **oo.** "Steering Committee" means the committee established in section 5.01 hereto;
- **pp.** "Steering Committee's Scope" means the scope, authority and business of the Steering Committee, as set out in section 5.03 hereto;
- **qq.** "Strategic Operating Committee" means the committee established in section 6.01 hereto:
- **rr.** "Strategic Operating Committee's Scope" means the scope, authority and business of the Strategic Operating Committee, as set out in section 6.01 hereto;
- **ss.** "Student" means (notwithstanding the University's definition of undergrad and grad student as set out in the University Calendar, and for the purposes of this Agreement) a full-time or part-time undergraduate or graduate student who is enrolled in at least one course (including, without restriction, thesis courses) for credit at the University of Alberta for a Fall Term, Spring Term, Summer Term or Winter Term, and is assessed a non-instructional fee for membership in the Students' Union or the GSA;

- tt. "Student Contributions Loan" means the loan described in section 3.01 (a) herein;
- **uu.** "Student Focused Areas" means the following spaces, located in either the PAW Centre or Other Renovated Areas:
 - i. Student Services Area, other than the Activity Registration Zone, Student/Faculty Pro Shop;
 - ii. Fitness Centre Area;
 - iii. Climbing Centre Area;
 - iv. Concourse Area; and
 - v. the Association Leased Space;
- vv. "Student Services Area" means all that portion of the Project designated or otherwise described as the Student Services area within the Design Development Report;
- ww. "Students' Council" means the council of elected representatives of the Students' Union;
- **xx.** "Students' Union" means the Students' Union of the University of Alberta;
- **yy. "Summer Term"** means each summer term as defined in the University's calendar, as amended from time to time;
- **zz.** "Term" has the meaning set out in section 11.01 hereto;
- aaa. "Users" means:
 - i. Students;
 - ii. Academic Staff and Non-Academic Staff; and
 - other individuals who are not Students, Academic Staff or Non-Academic Staff, who are permitted to use the Student Focused Areas by the University as contemplated in section 7.07 of this Agreement;
- **bbb.** "Winter Term" means each winter term as defined in the University's calendar, as amended from time to time; and
- **ccc.** "Work" means the planning, design, construction/renovation and commissioning of the PAW Centre and Other Renovated Areas.

SCHEDULE A-2 INTERPRETATION & GENERAL CONTRACTUAL TERMS

Construction and Interpretation

In the Agreement, including the Schedules, and in any amendments thereto, except as otherwise expressly provided, or unless the context otherwise requires:

- a. the recitals and headings to Articles, Sections and Schedules are for convenience only and will not affect the interpretation of the Agreement;
- **b.** all references in the Agreement or in any Schedule to "articles", "sections", "subsections", "paragraphs", "clauses", and "subclauses" or to other designated subdivisions are to the designated subdivisions of the Agreement or the applicable Schedule, as the case may be;
- each reference to a statute is deemed to be a reference to that statute and any successor statute, and to any regulations, rules, policies and criteria (to the extent binding and having the force of law) made under that statute and any successor statute, each as amended or re-enacted from time to time;
- **d.** each reference to a faculty, office, agency or similar body of any relevant authority is deemed to be a reference to any successor or replacement of such faculty, office, agency or similar body;
- each reference to a guideline, policy, regulation, rule or directive is deemed to be a reference to any successor or replacement of such guideline, policy, regulation, rule or directive;
- **f.** words importing the singular include the plural and vice versa, words importing gender include all genders, and words importing individuals shall include firms and corporations, and vice versa;
- g. references to time of day or date mean the local time or date in Edmonton, Alberta;
- **h.** all references to amounts of money mean lawful currency of Canada;
- i. the word "written" includes printed, typewritten, faxed, e-mailed or otherwise capable of being visibly reproduced at the point of reception and "in writing" has a corresponding meaning;
- **j.** the words "include" and "including" are to be construed as meaning "including, without limitation";
- **k.** except to the extent otherwise expressly provided by the Agreement, any requirement contained in the Agreement, including any Schedule, for the University to consult with, or have regard to the proposals or comments of any Person or to have due regard to, or take account of, any matter or representation will not be construed as a requirement to adopt, incorporate or comply with the result of any such consultation, such proposals or such matter;
- **l.** the words "herein" and "hereunder" and words of similar import refer to the Agreement as a whole including the Schedules and not to any particular section or other subdivision;
- **m.** any reference to an entity shall include and be deemed to be a reference to an entity (or entities) that is a successor, permitted assign or successor in title to such entity,

- including any entity which assumes by agreement, by operation of law or otherwise, the rights and/or obligations of the entity;
- **n.** words that have well-known technical or trade meanings and that are not specifically defined in the Agreement are used in the Agreement in accordance with their recognized meanings;
- any reference to "approval", "authorization" or "consent" of any Person, including any Party, means the written approval, written authorization or written consent of such Person;
- **p.** a reference to a day is a reference to a period of time commencing at midnight and ending the following midnight;
- **q.** a reference to a month is a reference to a calendar month;
- r. if the time for performing an obligation under the Agreement expires on a day that is not a Business Day, the time shall, unless specifically indicated to the contrary, be extended until that time on the next Business Day;
- **s.** where a word or phrase is specifically defined, other grammatical forms of that word or phrase have corresponding meanings; and
- t. a reference to time is a reference to the time in effect in Alberta, taking into account the *Daylight Saving Time Act* (Alberta).

Governing Law

The Agreement and each of the documents contemplated by or delivered under or in connection with the Agreement are governed exclusively by, and are to be enforced, construed and interpreted exclusively in accordance with, the laws of Alberta and the laws of Canada applicable in Alberta, which will be deemed to be the proper law of the Agreement.

Severability

If any provision of the Agreement is or becomes illegal, invalid or unenforceable, the illegality, invalidity or unenforceability of that provision will not affect the legality, validity or enforceability of the remaining provisions of the Agreement except that if:

- a. on the reasonable construction of the Agreement as a whole, the applicability of the other provision presumes the validity and enforceability of the particular provision, the other provision will be deemed also to be invalid or unenforceable;
- **b.** as a result of the determination by a court of competent jurisdiction that any part of the Agreement is unenforceable or invalid, the fundamental purpose of this Agreement is frustrated, this Agreement shall be either renegotiated or terminated by the Parties.

Further Assurances

The Parties shall, with reasonable diligence, hold all meetings, perform all acts, execute and deliver all documents and instruments, do all things and provide all reasonable assurances as may be reasonably necessary or desirable to give effect to the provisions and intent of the Agreement and to complete the transactions contemplated by the Agreement.

Approvals

Save and except for as specifically stated otherwise within this Agreement (including, without restriction, where stated to be within the discretion or sole discretion of a Party), no approval, authorization, sanction or permission required to be provided under the Agreement shall be unreasonably or arbitrarily withheld or delayed by the Party providing same.

Waivers

No waiver of any provision of the Agreement is binding unless it is in writing and signed by all the Parties, except that any provision which does not give rights or benefits to particular Parties may be waived in writing, signed only by those Parties who have rights under, or hold the benefit of, the provision being waived if those Parties promptly send a copy of the executed waiver to all other Parties. No failure to exercise, and no delay in exercising, any right or remedy under the Agreement will be deemed to be a waiver of that right or remedy. No waiver of any breach of any provision of the Agreement will be deemed to be a waiver of any subsequent breach of that provision or of any similar provision.

Enurement

The Agreement shall enure to the benefit of and be binding upon the Parties and their respective successors and permitted assigns.

Exclusive Warranties

The warranties expressly set forth in the Agreement are exclusive and are in lieu of all other warranties, whether statutory, express or implied (including warranties of merchantability and fitness for a particular purpose, and warranties arising from course of dealing or usage of trade).

Counterparts

The Agreement and all documents contemplated by or delivered under or in connection with the Agreement may be executed and delivered in any number of counterparts with the same effect as if all Parties had signed and delivered the same document and all counterparts will be construed together to be an original and will constitute one and the same agreement.

Delivery by Fax

Any Party may deliver an executed copy of the Agreement by facsimile or electronic transmission provided that the Party will promptly dispatch by delivery in person to the other Party an originally executed copy of the Agreement.

Amendments

Except as specifically provided in the Agreement, no amendments, supplement, restatement or termination of any provision of the Agreement is binding unless it is in writing and signed by each Party at the time of the amendment, supplement restatement or termination.

Submission to Jurisdiction

Subject to the Conflict and Dispute Resolution Procedure, each of the Parties irrevocably submits to the exclusive jurisdiction of the courts of Alberta and all courts having appellate jurisdiction over those courts in any suit, action or other proceeding in any way related to or arising out of the Agreement by any Party against the other Party.

Entire Agreement

The Agreement, including the attached schedules, and all documents contemplated by or delivered under or in connection with the Agreement, constitutes the entire agreement between the Parties with respect to the subject matter of the Agreement and supersede all prior agreements, negotiations, discussions, undertakings representations, warranties and understandings in respect thereof, whether written or oral, express or implied, statutory or otherwise

Continuing Nature of Rights and Obligations

The expiry or termination of the Agreement shall not relieve any Party of any rights, liabilities or obligations that by their nature survive expiry or termination, including warranties, remedies, indemnities and obligations of confidentiality or environmental compliance, or that arose prior to the expiry or termination of the Agreement.

No Application of Contra Proferentem

The provisions of the Agreement were negotiated by the Parties and the Agreement shall be deemed to have been drafted by both Parties. The Parties hereby acknowledge that they have read the Agreement in its entirety and that each has obtained independent legal advice in connection with the preparation and execution of the Agreement and the principle of *contra proferentem* shall not be argued, pleaded or applied in any proceeding subsequent to the execution hereof in respect to any provision of the Agreement.

SCHEDULE A-3 REFERENDUM QUESTIONS

In March of 2010, undergraduate students passed the following referendum:

Do you support the establishment of a non-instructional fee subject to the following conditions to build and operate a Physical Activity and Wellness (PAW) Centre on North Campus, subject to the following conditions?

- 1. The fee would be dedicated to the construction of a new Physical Activity and Wellness (PAW) Centre on North Campus as well as concurrent upgrades to the Van Vliet Physical Education and Recreation Centre.
- 2. The fee would be assessed in each term to undergraduate students enrolled in at least one course for credit, subject to the following conditions:
 - a. The fee shall be no greater than \$29.00 per Fall and Winter Terms;
 - b. The fee shall be no greater than \$14.50 per Spring and Summer Terms;
 - c. The fee shall be optional for:
 - i. Students that are registered only in courses designated as off-campus;
 - ii. Students that are registered only in courses at Augustana campus;
 - d. The fee shall provide unrestricted use of the Physical Activity and Wellness Centre during its hours of operation to each undergraduate student currently paying the fee; and
 - e. The fee shall not be assessed until the construction of the facility is complete and the student component is operational;
 - f. The fee shall be assessed for a period no longer than 35 years, or until the costs incurred in (1) have been recovered; and
 - g. Undergraduate students would be unable to alter or rescind the fee.
- 3. There shall be a committee established for the Physical Activity and Wellness Centre that shall:
 - a. Have an overall student majority;
 - b. Provide strategic direction for the student component of the building program; and
 - c. Establish budget principles annually.
- 4. A building agreement is established between the Governors of the University of Alberta and the University of Alberta Students' Union that is subject to approval by Students' Council.

In March of 2010, graduate students passed the following referendum:

Do you agree to the establishment of a new mandatory fee to build and operate a Physical Activity and Wellness (PAW) Centre on North Campus, subject to the following conditions?

- 1. The fee will be dedicated to the construction of a new Physical Activity and Wellness (PAW) Centre on North Campus as well as concurrent upgrades to the Van Vliet Physical Education and Recreation Centre.
- 2. Students shall be assessed a fee no greater than \$29.00 every four months subject to the following conditions:
 - a. The fee shall provide unrestricted use of the Physical Activity and Wellness Centre during its hours of operation to each graduate student currently paying the fee.
 - b. The fee shall not be assessed until the construction of the facility is complete and the student component is operational;
 - c. The fee will be optional for students enrolled at Augustana Campus and students registered in off-campus thesis;
 - d. Thefeeshallbeassessedforaperiodnolongerthan35years,or until the costs incurred in (1) have been recovered, whichever happens first; and
 - e. Graduate students would be unable to alter or rescind the fee.
- 3. There shall be a committee established for the Physical Activity and Wellness Centre that shall:
 - a. have an overall student majority,
 - b. provide strategic direction for the student component of the building program, and
 - c. establish budget principles annually.
- 4. A building agreement is established between the Governors of the University of Alberta, the Students' Union of the University of Alberta and the Graduate Students' Association of the University of Alberta that is subject to approval by GSA Council.

SCHEDULE B-1 FUNCTIONAL REQUIREMENTS

In conducting the design, planning and construction of the Project in accordance with this Agreement, the University shall comply with and deliver upon the following functional requirements:

1. The University shall:

- (a) complete the Project substantially in accordance with the plans and specifications in the Design Development Report, as aspects of such Design Development Report are modified and subsequent plans and specifications are developed in accordance with this Schedule B-1. For clarification, notwithstanding the foregoing the Project shall be deemed to be non-compliant with the foregoing in the event that:
 - (i) the University eliminates in its entirety any functional space/program area comprising the Student Focused Areas, as contemplated within Section 2.4 of the Detailed Design Report, from the Project delivered on the Acceptance Date without the prior written agreement of the Associations;
 - (ii) the University makes any material change to the Student Focused Areas for the purpose of permitting the Student Focused Areas to be completed at a cost that is not more than the maximum permitted principal amount of the Student Contributions Loan, without first:
 - (1) if feasible, providing a proposal for an alternative means to deliver the functional/program area within the Project, which proposal may consider an alternative timeline for completing the Student Focused Areas; and
 - consulting with the Steering Committee regarding the alternative proposal;

Subject to the provisions in this Schedule, modifications to the size, layout, configuration, or specifications of the Student Focused Areas or Other Renovated Areas shall not, in and of themselves, cause a non-compliance with the obligation to complete the Project substantially in accordance with the plans and specifications in the Design Development Report;

(b) take reasonable steps to commence construction of the Project within twenty four (24) months of the effective date of this Agreement unless otherwise agreed to be extended by the Parties;

- (c) diligently pursue completion of the Student Focused Areas after construction of the Project has commenced;
- (d) as part of completing the Project, install appropriate equipment in the Student Focused Areas, other than the Association Leased Space, to facilitate the use of such areas for their intended purposes. Without limiting the generality of the foregoing, in the Fitness Centre Area, the University shall install:
 - (i) cardiovascular equipment such as treadmills, ellipticals, summit trainers, stationary bikes, steppers, rowers, and ergometers;
 - (ii) selectorized (machine) equipment;
 - (iii) plate load equipment;
 - (iv) benches and racks;
 - (v) free weights and plates;
 - (vi) functional equipment

and other fitness equipment, the details of which shall be discussed by the Steering Committee.

- 2. The Steering Committee shall meet as required in order to discuss any amendments that may be required in the course of implementation of the Design Development Report; the design and construction shall be informed by various matters including program operations, building operations and costing exercises.
- 3. The University may make changes to the Project in accordance with the normal University processes subject to the provisions of paragraphs 4 and 5 of this Schedule.
- 4. If a material negative change to the Student Focused Areas is proposed other than for the purpose of permitting the Student Focused Areas to be completed at a cost that is not more than the maximum permitted principal amount of the Student Contributions Loan, and the President (or designate) of the Student's Union or the President (or designate) of the GSA declares at the Steering Committee meeting at which a decision is to be made on such proposal, or provides written notice within Five (5) Business Days following the date of such meeting, that such proposal is not acceptable to that President's (or designate's) Association, the University shall not implement such change unless the same is subsequently approved in writing on behalf of that Association.
- 5. For the purposes of section 4 of this Schedule B-1, a material negative change means a change to the design or specifications of the Student Focused Areas that, individually or in the aggregate, is material and adverse to the use and enjoyment of the PAW Centre by the Students or the Associations, or would prevent or materially impair the ability of the Students or the Associations to utilize the impacted portion of the Student Focused Areas for the purposes for which they were originally intended.
- **6.** If the University and either Association disagree as to:

- (a) whether a proposed change to the Student Focused Areas is a material negative change; or
- (b) whether a material negative change to the Student Focused Areas is required for the purpose of permitting the Student Focused Areas to be completed at a cost that is not more than the maximum permitted principal amount of the Student Contributions Loan;

such disagreement shall be treated as a Conflict and shall be resolved in accordance with the Conflict resolution procedure set forth in this Agreement.

7. The University shall, endeavour and utilize reasonable efforts to incorporate visual demonstrations of sustainable construction and operation into the final design and specifications of the project working within the framework of a sustainable certification process following Green Globes design, environmental assessment for new buildings, and shall use all reasonable efforts to seek a minimum certification level of 3 Globes (which the parties understand to be equivalent to a LEED silver). If possible within budget, scope limits and site constraints of the project the University will seek to achieve through a variety of potential features, specifications and design elements forming part of the work a level of 4 Globes (which the parties understand to be equivalent to a LEED gold).

SCHEDULE C-1 OPERATIONAL REQUIREMENTS

In conducting the Operation and Maintenance of the Project in accordance with this Agreement, the University shall comply with and deliver upon the following operational requirements for, in respect of, or otherwise affecting the Student Focused Areas:

- 1. **Objectives** In providing services with respect to the management, operation, maintenance and repair of the Common Areas and the Student Focused Areas, other than Association Leased Space, the University shall:
 - **a.** preserve the integrity, value, and functional ability of the physical assets in a reasonable and prudent manner;
 - **b.** deliver the services in a well-planned, high quality, and cost effective manner; and
 - **c.** continuously meet the spirit and intent of the Agreement Principles.
- 2. Service Standards Notwithstanding anything contained within this Agreement, the University shall provide Operation and Maintenance of the Common Areas and the Student Focused Areas, other than Association Leased Space, throughout the Term in such a manner so as to meet the following service standards:

a. General:

- the University shall use all reasonable efforts to achieve and maintain BOMA BESt Level 3 certification, or its agreed upon equivalent (as may be current from time to time throughout the Term) in the delivery of Operation and Maintenance;
- ii. the University shall use all reasonable efforts to maintain or cause to be maintained all features, specifications and design elements which were incorporated into the construction and specification of the Project and the equipment and services utilized therein so as to continue to attain the Green Globes standard achieved by the Project following the completion of the construction of the Project; and
- iii. subject to the following specific operational requirements, the Common Areas and the Student Focused Areas, other than the Association Leased Space, shall be operated and maintained by the University in a manner consistent with other buildings operated and maintained by the University and in substantial conformance with the University's general policies, with due regard to their intended use, and their purpose of health and wellness.

b. Specific:

- i. Subject to the Agreement Principles:
 - **A.** The University will use reasonable efforts to maintain operating hours of the Fitness Centre Area substantially similar to the following:

Hours from September through April inclusive: 6:30 am to 10:00 pm Monday through Friday 8:00 am to 8:00 pm Saturday 9:00 am to 8:00 pm Sunday

Hours from May through August inclusive: 6:30 am to 10:00 pm Monday through Friday 8:00 am to 7:00 pm Saturday 9:00 am to 6:00 pm Sunday

subject to either shorter hours or closure from and including Christmas Day to and including New Years Day, on statutory and civic holidays, and other University-wide building closures. Nothing in this Agreement precludes the University from maintaining operating hours in excess of the hours set out above;

- **B.** The Fitness Centre may be closed on a temporary basis as may reasonably be required due to maintenance or repairs, or due to emergency;
- C. Subject to any other applicable provisions of the body of the Agreement of which this Schedule forms a part, the Fitness Centre shall be available to Students during all hours of operation without requirement for additional fee (not including participation in programs or activities, which may be subject to additional fees);
- **D.** At no time shall the Fitness Area be open for operation and use by some Users to the exclusion of some or all Students (subject to the University's ability to exclude certain Students in accordance with the *Code of Student Behaviour*); and
- **E.** If the Strategic Operating Committee recommends that the University change the hours of operation for the Fitness Centre, then the University may do so.
- ii. Notwithstanding the foregoing, but always subject to the Agreement Principles, the Climbing Centre Area shall:

- **A.** be subject to reasonable safeguards in relation to wall access, as determined by the University, to address risk management concerns relative to safe use and facility operation;
- **B.** be available to Students without the requirement of any additional fee on a drop in, first come first served basis during all hours in which the Climbing Centre is open for operation and use by any Users, and at no time shall the Climbing Centre Area be open for operation and use by some Users to the exclusion of some or all Students (subject to the University's ability to exclude certain Students in accordance with the *Code of Student Behaviour*);
- C. subject to any other applicable provisions of the body of the Agreement of which this Schedule forms a part, be open for operation and use during the following times:

September – April

Monday to Friday 11:00-16:00 Saturday 12:00-16:00

May – August

Monday to Friday 16:00-19:00

Nothing in this Agreement precludes the University from maintaining operating hours in excess of the hours set out above. If the University determines that sufficient demand exists, it may also make the Climbing Centre Area open for operation and use during additional periods such as on Saturdays during the May – August period, and Sundays.

Such access is subject to the following:

- 1. Subject always to due consultation with the Strategic Operating Committee, the hours in which the Climbing Centre Area shall be open for operation and use by all Users may be shortened or may be shifted to be earlier or later in the day;
- 2. Such operating hours may be shortened, or the Climbing Centre Area may be closed, during the period of time from and including Christmas Day to and including New Years Day, on statutory and civic holidays, and during other University-wide building closures;
- 3. The Climbing Centre Area may be closed on a temporary basis as may be reasonably required due to maintenance or repairs, or due to emergency. Any proposed significant disruptions of the schedule for the Climbing Centre Area (for example, annual maintenance, or restricted access to accommodate a major climbing competition)

- will be taken to the Strategic Operating Committee for information and feedback;
- **4.** Only certain sections of the bouldering and high wall in the Climbing Centre Area may be open at a particular time;
- 5. Use of the Climbing Centre Area may not be exclusively for Students during the hours in which the Climbing Centre Area is open for operation and use;
- 6. During selected time periods only, the University may implement a reservation fee to allow Users to reserve climbing time in the Climbing Centre Area. During these time periods, at least 50% of climbing space shall continue to be made available on a drop-in, first-come, first-serve basis. The Strategic Operating Committee shall be consulted on the applicable time periods, the frequency of the time periods, and any reservation fees prior to implementation of the reservation system and fee;
- 7. Use of the zones within the Climbing Centre Area will be planned to maximize participation and cost effectiveness;
- 8. The Student entitlement to access to and use of the Climbing Centre Area includes access to the climbing walls and, where applicable, top ropes, but does not include free use of climbing shoes, harnesses, and other personal safety equipment, or additional movable equipment. The University may charge reasonable rental fees for such equipment, and the Strategic Operating Committee shall be consulted on such fees;
- 9. All Students will pay for the annual access re-certification (i.e. top rope test or lead test) prior to doing lead climbing or high wall climbing within the Climbing Centre Area as is imposed for all Users (other than Users under supervised extra-fee lesson programs);
- 10. The Parties expect that patterns of use and student needs in the Climbing Centre Area will need to be established and evaluated over the course of the first year or two so that the University can ensure the Climbing Centre Area is running effectively and in a fiscally responsible manner, and it can establish the appropriate pattern of usage for establishing the hours in which the Climbing Centre Area is open for operation and use;
- 11. It is anticipated that the ongoing discussion relative to the Climbing Centre Area will be taken to the Strategic Operating Committee so

- that the Strategic Operating Committee can provide response and input to proposals from the University relative to the hours in which the Climbing Centre Area is open for operation and use;
- 12. The Strategic Operating Committee will be able to provide its proposals or concerns to the University so that needs and discussion items can be addressed jointly;
- **13.** Fees charged in association with Climbing Centre Area and related equipment usage shall be utilized for Climbing Centre Area operational and capital costs; and
- **14.** If the Strategic Operating Committee recommends that the University change the hours of operation in which the Climbing Centre Area is open for operation and use, then the University may do so.

At no time shall the access to and use of the Climbing Centre Area by Students as provided for within Section 2(b)(ii)(B) above be amended or otherwise restricted by the University or any operator of the Climbing Centre Area without the prior written agreement of the Associations or without regard to the principles laid out in the Referenda attached as Schedule A-3 hereto.

SCHEDULE D-1 FORM OF LEASE(S)

THIS AGREEMENT	made as	of the	day of		2011.
----------------	---------	--------	--------	--	-------

BETWEEN:

THE GOVERNORS OF THE UNIVERSITY OF ALBERTA (the "University")

-andTHE STUDENTS' UNION OF THE UNIVERSITY OF ALBERTA
(the "Students' Union")
-and-

THE GRADUATE STUDENTS' ASSOCIATION OF THE UNIVERSITY OF ALBERTA (the "GSA")

PHYSICAL ACTIVITY AND WELLNESS (PAW) CENTRE LEASE AGREEMENT

WHEREAS:

- A. The parties have entered into the PAW Centre Project Agreement which sets out the principles governing the establishment of the PAW Centre, and their respective obligations and contributions to the planning, design, construction/renovation, commissioning, funding, maintenance and ongoing operations of the PAW Centre and certain areas within the Van Vliet Centre;
- **B.** The PAW Centre Project Agreement provides that the University shall lease to the Tenant the Premises during the Term on the terms and conditions set forth in this Agreement.

NOW THEREFORE in consideration of the above and other good and valuable consideration, the receipt, sufficiency and benefit of which are hereby acknowledged, each of the University and the Tenant agrees with the other as follows:

Definitions and Schedules

1.08 Definitions

In this Lease, unless something in the subject matter or context is inconsistent therewith:

- **a.** "Acceptance Date" has the meaning attributed to such term in Section 3.04 of the PAW Centre Project Agreement;
- **b.** "Building" means such of the PAW Centre or the Van Vliet Centre in which the Premises are situate;
- **c. "Common Use Areas"** means those areas within the Building not intended for exclusive use of the University, the Tenant or any other tenant or authorized occupant of the University as designated from time to time by the University and set aside for the

common or joint use of the occupants of the Building in the manner and for the purposes for which the same are so designated and if the Building is the PAW Centre does not include the Student Focused Areas or the Academic and Research Areas as each of such terms is defined in the PAW Centre Project Agreement;

- **d.** "Lease" or "Agreement" means this Physical Activity and Wellness Centre Lease which includes all schedules hereto;
- **e. "PAW Centre"** means the new building so designated and to be constructed in accordance with the PAW Centre Project Agreement;
- **f. "PAW Centre Project Agreement"** means the Physical Activity and Wellness Centre Project Agreement made between the parties concurrently herewith;
- **g. "Premises"** means the space outlined in red on Schedule "A" hereto, the boundaries of which shall extend from the top surface of the structural subfloor to the bottom surface of the structural ceiling or the building standard ceiling finish (whichever is lower), shall extend to the unfinished surfaces of all walls which bound all or any portions of the Premises, and shall exclude any structural components of the Building;
- **h.** "Tenant" means collectively the Students' Union and the GSA;
- **i. "Tenant Improvements"** means any alterations, renovations, changes, improvements and additions made to the Premises by or on behalf of the Tenant or any person claiming through or under the Tenant including any fixtures or equipment attached in any way to the Building but excludes the Work;
- **i. "Tenant Property"** means all of the personal property located in the Premises, other than any personal property owned by the University, including but not limited to improvements, equipment, stock-in-trade, furniture, fixtures and signs whether owned by, in the custody of or permitted to be in the Premises by the Tenant or any person claiming through the Tenant but excluding Tenant Improvements;
- **k.** "**Term**" means the period commencing as of the Acceptance Date and terminating on the earliest of:
 - i. the date of termination of the PAW Centre Project Agreement; and
 - ii. such other date as determined pursuant to this Lease.
- **l. "Van Vliet Centre"** means the building on the campus of the University known as the Van Vliet Physical Education and Recreation Centre.
- **m.** "Work" means the planning, design, construction/renovation and commissioning of the PAW Centre and the Other Renovated Areas (as Other Renovated Areas are defined in

the PAW Centre Project Agreement) in accordance with the PAW Centre Project Agreement.

Grant and Acceptance of Lease

2.01 The University leases to the Tenant and the Tenant accepts from the University such lease of the Premises for the Term on the terms and conditions contained herein.

2.02 This Lease:

- **a.** is entered into pursuant to the PAW Centre Project Agreement;
- **b.** applies only with respect to the relationship of landlord and tenant in relation to the Premises that being the sole relationship between the parties established pursuant to this Lease;
- does not affect the terms and conditions contained in the PAW Centre Project Agreement with respect to any other relationship between the University and the Students' Union and the GSA or the relationship of landlord and tenant between those parties as provided in the PAW Centre Project Agreement with respect to space other than the Premises;
- d. together with any relevant provision of the PAW Centre Project Agreement constitutes the whole agreement of the University and the Tenant with respect to the Premises provided that where there is conflict between any term or condition contained in this Lease and any term or condition contained in the PAW Centre Project Agreement the term or condition contained in this Lease governs.

Payments to University

- 3.01 The University acknowledges receipt and the sufficiency of the sum of \$10 and other good and valuable consideration from the Tenant as provided in the PAW Centre Project Agreement as full payment of basic annual rent (the "Annual Rent") for the Term.
- 3.02 The Tenant shall pay to the University such amounts as required pursuant to this Lease to the University at the office of the University hereinafter set forth or at such other place as the University may in writing, from time to time, direct without notice or demand unless otherwise provided in this Lease.
- 3.03 The Tenant shall pay the University interest at a rate equal to the prime bank lending rate from time to time charged by the University's main bank in Edmonton on short term loans in Canadian dollars plus Three (3%) percent per annum on all arrears of rent or other sums payable by the Tenant to the University herein from the date of default in payment, or where provided herein, from the date the amount was expended or incurred by the University, until payment is received by the University.

Use

- 4.01 The Tenant may use or sub-let the Premises for commercial business activities that provide of support services to students as defined in the regulation to the Post-Secondary Learning Act (Alberta) as amended from time to time and for no other purposes and the Tenant shall not carry on any business, occupation or activities or permit anything to be carried on or done on the Premises which shall be deemed a nuisance or that shall cause annoyance to other occupants of the Building or by which any insurance on the Premises or the Building or any part thereof may become void or voidable.
- 4.02 The University has entered into and will continue to enter into exclusive campus-wide sponsorship, advertising or supply arrangements for food and beverages and vending machines pursuant to which restrictions are imposed on the activities of the occupants of premises on the campus of the University and pursuant to which the University has an obligation to observe such restrictions and to have each tenant covenant to observe such restrictions.

The University shall ensure that it does not enter into future campus wide sponsorship, advertising or supply arrangements affecting the Premises, without first consulting with the Tenant. The Tenant shall comply with restrictions on its use and occupation of the Premises required by those arrangements that are have been established by the University prior to the date of this Agreement. The Tenant shall also comply with restrictions on its use and occupation of the Premises required by those arrangements that are, in accordance with this Section, entered into by the University from and after the date of this Agreement, provided those restrictions do not cause significant material negative impacts on the business being undertaken by the Tenant or by a sub-tenant or a licensee of the Tenant within the PAW Centre or the Van Vliet Centre at the time that the University enters into such arrangement.

4.03 a. The Tenant shall not:

- i. commence to use:
- ii. enter into any agreement permitting any other person, including any sub-tenant of the Tenant the right to commence to use; or
- **iii.** grant to any other person under any existing agreement a consent, approval or permission, where it is otherwise the right of the Tenant under such agreement to withhold such consent, approval or permission, to use;

any portion of the Premises for a commercial business activity which business activity would compete or interfere with any commercial business activity being undertaken at the time of commencement of such use by the University or by a tenant or a licensee of the University within the PAW Centre or the Van Vliet Centre.

b. In addition to the restriction contained in Section 4.03(a), during the period of time in which the University permits the existing Booster Juice business to continue to operate in the Van Vliet Centre, the Tenant shall not commence to use, grant to any sub-tenant or

licensee of the Tenant a right to use, or permit any sub-tenant or licensee of the Tenant to commence to use, any portion of the Premises for the sale or provision of fruit based smoothies or fruit based juices other than the sale or provision of pre-packaged juices the sale or provision of which otherwise complies with any applicable arrangements referred to in Section 4.02 and any other applicable provisions of this Lease.

- **4.04** The University shall not, except as otherwise provided in Section 4.02:
 - a. commence to use;
 - **b.** enter into any agreement permitting any other person, including any tenant of the University the right to commence to use; or
 - c. grant to any other person under any existing agreement a consent, approval or permission, where it otherwise the right of the University under such agreement to withhold such consent, approval or permission, to use;

any portion of the PAW Centre or the Van Vliet Centre for a commercial business activity which business activity would compete or interfere with any commercial business activity being undertaken at the time of commencement of such use by the Tenant or by a sub-tenant or a licensee of the Tenant within the PAW Centre or the Van Vliet Centre.

4.05 Each of the University and the Tenant shall in commencing to use, entering into an agreement permitting any other person to use or granting to any person under any existing agreement a consent, approval or permission to use space referred to in Section 4.03(a) or Section 4.04, as the case may be, for commercial business activity consider the manner in which such use contributes to the Mission of the PAW Centre, as defined and contemplated in the PAW Centre Project Agreement. At the request of either the University or the Tenant the Strategic Operating Committee may review any action or proposed action by either the University or the Tenant that the requesting party believes may be in contravention of the provisions of Section 4.03(a) or Section 4.04. If either the Landlord or the Tenant disagrees with the results of that review or with the reviewed action or proposed action of the other that disagreement shall be treated as a conflict pursuant to the PAW Centre Project Agreement and shall be resolved in accordance with Schedule F-1 (Conflict and Dispute Resolution Procedure) of such agreement.

Telephone, Taxes and Other Charges Payable by Tenant

5.01 The University shall provide telephone line service and data communication line service to the Premises as provided in the Work. Any additional telephone line service and data communication line service within the Premise shall be the responsibility of the Tenant. The Tenant shall arrange for and pay for any and all required telephone and telecommunication installation and use charges. All arrangements for telephone services or equipment on University property must be coordinated through the University's Academic Information and Communication Technology.

- 5.02 The Tenant shall pay on or before their respective due dates all other Tenant's charges in the nature of business taxes, property taxes and other Tenant's charges of a similar nature of a tax that may be assessed or charged against the Tenant by a valid taxing authority with respect to its operations on or from the Premises.
- 5.03 The Tenant shall pay to the University the cost of any and all utilities, including those referred to in Section 9.01 which are provided to the Premises by the University and consumed upon the Premises which cost shall if there are separate meters or check-meters for the Premises be based on the consumption disclosed by such meters and otherwise shall be a portion of the cost of the Utilities consumed within the Building which portion is equal to the portion, as calculated by the University's Office of Facilities and Operations, that the area of the Premises is of the total area of the Building.
- 5.04 The Tenant shall pay to the University the cost incurred by the University for the janitorial services for the Premises as contemplated in Section 9.01. Unless the janitorial services are provided pursuant to a janitorial contract between the University and a third party and there is a separate cost allocable to the Premises the cost incurred by the University for the janitorial services for the Premises shall be determined by the University's Office of Facilities and Operations and shall be the portion of the total costs of janitorial service for the Building that the area of the Premises is of the total area of the Building. Alternatively, the Tenant may choose to provide its own janitorial services for the Premises.
- 5.05 The costs payable by the Tenant pursuant to each of Section 5.04 and Section 5.05 shall be paid within thirty (30) days of monthly invoice by the University.
- 5.06 The Tenant shall pay to the University on the day the same fall due from the Tenant under the legislation applicable thereto any and all tax required to be collected from the Tenant by the University and known as Goods and Services Tax.

Compliance with Laws, Regulations and Rules

- 6.01 The Tenant shall observe, fulfill and comply with the provisions and requirements of any and all laws, orders, regulations, by-laws and ordinances of governmental bodies relating to the use and occupancy of the Premises by the Tenant, and without limiting the generality of the foregoing shall, in accordance with the provisions of this Lease, make such alterations and changes to the Premises required to make the same comply with the foregoing.
- 6.02 The Tenant shall observe, fulfill and comply with, and shall cause its employees to observe, fulfill and comply with, all rules and regulations from time to time of the University relating to the use of the Premises and the Common Use Areas in the Building, provided the same are not inconsistent with the terms of this Lease.

Repairs

- 7.01 Except as otherwise provided in this Lease, the Tenant shall maintain the Premises, including the Tenant Improvements in good condition during the term and shall make at its sole cost and expense, all necessary repairs and replacements thereto, except for normal wear and tear and repairs and replacements for which the University is responsible under the provisions of this Lease. Such repairs and replacements and any work carried out by the Tenant as contemplated in Section 8.01 shall be made or carried out at the option of the University only by, or under the direct supervision and control of, the University's Office of Facilities and Operations, shall be carried out by skilled and reputable contractors acceptable to the University and shall be carried out with materials of a good and substantial quality.
- 7.02 The Tenant shall permit the University, its servants, employees and agents to enter upon the Premises at all reasonable times for the purpose of viewing the condition thereof and shall within ten (10) days of receipt of written notice given by or on behalf of the University commence and proceed diligently with the execution of such repairs and work required to be done by it pursuant to this Lease and mentioned in such notice. In the event of the Tenant's failure or neglect to carry out any such repairs or work, the University and its agents may enter the Premises and at the Tenant's expense, perform and carry out such repairs and work and the University shall not be liable for any inconvenience, disturbance, loss of business or other damage resulting therefrom and in the event the University spends any monies pursuant to this Section the Tenant will pay the same on demand together with interest from the date of such expenditure by the University.
- 7.03 Subject to the provisions of Section 13.01, the Tenant shall, upon termination of the term herein granted or any renewal thereof, leave the Premises in the same condition as it is required to maintain the same throughout the term and shall remove from the Premises any signs installed by the Tenant after the commencement of the term of this Lease and required to be so removed by the University and shall restore those parts of the Premises to which those signs were affixed to the same condition as existed at the commencement of the Lease
- 7.04 The Tenant shall give prompt written notice to the University of any defect, damage, disrepair or malfunction of any structural portion of the Building or any part of the heating, plumbing, electrical, ventilating, air conditioning or lighting systems in the Building.
- **7.05** The University shall, from time to time, throughout the term:
 - a. carry out as soon as reasonably possible in the circumstances after receipt of notice thereof in writing from the Tenant, structural repairs to the foundations, exterior walls, structural sub-floors, the roof, the structural portions of bearing walls and structural columns and beams which if not carried out would interfere with or impair the use, occupancy or safety of the Premises;
 - **b.** carry out as soon as reasonably possible in the circumstances after receipt of notice thereof in writing from the Tenant, repairs to the electrical, mechanical, plumbing and

- utilities systems within or serving the Premises to the extent such systems do not form a part of Tenant Improvements or Tenant Property;
- subject to Section 13.01, repair all damage to the Premises other than any Tenant Improvement or Tenant Property which is covered by any insurance effected by the University in accordance with the provisions of Section 10.01 to the extent of the proceeds of such insurance applicable thereto and received by the University.

Alterations and Fixturing

- 8.01 The Tenant shall not make or install any Tenant Improvements or install any fixtures or equipment in the Premises that require attachment in any way to the Building without the consent of the University first had and obtained, which consent shall not be unreasonably withheld. All such work undertaken by the Tenant must be coordinated and carried out through the University's Office of Facilities and Operations. All alterations, improvements, finishes and renovations shall immediately become part of the Premises and the Building, provided always that except as otherwise agreed in writing the University may at the expiration or sooner termination of this Lease require the Tenant restore the Premises in whole or part to the same condition in which they were at the time of entering into this Lease.
- 8.02 The University shall complete the Work, if any, in a good and workmanlike manner and shall use its reasonable efforts to complete that Work in accordance with the time schedule established by the University for the completion thereof. However, the University shall not be liable for loss, injury, damage or inconvenience which the Tenant may sustain by reason of the inability or failure of the University to complete such work and/or deliver possession of the Premises in accordance with that time schedule. The University shall at all times have access to the Premises for the conduct of the Work.
- **8.03** The Tenant acknowledges that as at the Acceptance Date, the Premises are in a good and tenantable state of repair and the Work, if any, in relation to the Premises has been completed in a satisfactory manner, subject always to those defects or deficiencies which are identified in a written list prepared by the University and Tenant in a jointly undertaken turn-over inspection of the Premises prior to the Acceptance Date.

Maintenance and Other Services

9.01 The University shall supply electrical power, water, natural gas and any other utilities distribution systems for which are included in the Work to the Premises and shall provide janitorial services for the Premises. The provisions of this Section 9.01 shall not oblige the University to provide electrical power, water, natural gas or any other utilities or janitorial services to the Premises in excess of the standards therefore which are generally applicable to other space within the Building.

- 9.02 The Tenant shall, at its expense and in accordance with the approvals of the University through University of Alberta Protective Services ("UAPS") (780 492-5050) and the University's Office of Facilities and Operations, make all necessary arrangements for securing the Premises, such arrangements to include access rights for UAPS, the University's Office of Facilities and Operations, and any other properly authorized persons in the event of emergency or for the performance of the University's obligations or the exercise of its rights pursuant to the Lease.
- **9.03** The Tenant acknowledges that this agreement is for use of space only. The Tenant must make arrangements with the appropriate department for provision of additional university services and must establish methods for payment for said services with the department of Financial Services.

Insurance

- 10.01 The University shall at all times during the Term maintain at its sole cost all risk insurance in respect of the Building, excluding any Tenant Improvements and Tenant Property, for the full replacement value thereof provided that such insurance, without further consent of, or notice to, the Tenant may have such deductible as the University may determine. Such all risk insurance shall contain a waiver of all rights of subrogation of the insurer against the Tenant, its agents, employees and those other persons for whose actions the University is in law responsible The University may, but shall not be obligated to, carry such other insurance including comprehensive general liability insurance related to the Building or the rents issuing therefrom or such risks and perils in relation thereto or the University's interest therein as the University may determine.
- 10.02 Every right, exemption from liability, defense and immunity of whatsoever nature applicable to the Tenant or to which the Tenant is entitled pursuant to the provisions of Section 10.01 shall also be available and shall extend to protect each agent, employee and other person referred to in Section 10.01 against whom subrogation is waived and for the purpose of all the foregoing provisions of this Section 10.02, the Tenant is or shall be deemed to be acting as agent or trustee on behalf of and for the benefit of each person who is or who becomes such agent, employee or other person from time to time.
- 10.03 The Tenant shall at all times during the Term maintain at its sole cost the following insurance:
 - a. general public liability insurance against claims for personal injury, death or property damage occurring upon, in or about the Premises, such insurance to afford protection to the combined single limit for personal injury, death or property damage of five million (\$5,000,000) Dollars or such other limits as may be approved or requested by the University;
 - **b.** plate glass insurance to the full insurable value of all plate glass on the Premises;
 - **c.** all risk insurance on the Tenant Improvements and the Tenant Property to the full replacement value thereof;

- **d.** boiler and machinery insurance on such boilers and pressure vessels as may be installed by the Tenant in the Premises;
- **e.** Tenant's legal liability insurance with respect to the Premises with a limit of not less than the replacement value of the Premises;
- **f.** such other risks and liabilities and for such amounts the University may reasonably request.

The policies of insurance referred to above shall have the following provisions or characteristics:

- g. any loss which would otherwise be payable to the University shall be payable to the University notwithstanding any act or neglect of the Tenant which might otherwise result in the forfeiture of such policies or any of them;
- h. the same shall not be affected or invalidated by any act, omission or negligence of any third party which is not within the knowledge or control of the insured;
- i. all policies of public liability insurance shall be written to cover the University and the Tenant and shall provide that each person firm or corporation insured under such policy or policies shall be insured in the same manner and to the same extent as if individual policies had been issued to each of them;
- **j.** such policies of insurance shall not be cancelled without at least thirty (30) days prior written notice to the University from the insurer;
- **k.** all such policies of insurance, other than policies of public liability insurance, shall contain a waiver of all rights of subrogation of the insurer against the University, its agents, employees and those other persons for whose actions the University is in law responsible.
- 10.04 Every right, exemption from liability, defense and immunity of whatsoever nature applicable to the University or to which the University is entitled pursuant to the provisions of Section 10.03 and pursuant to Section 10.05 shall also be available and shall extend to protect each agent, employee and other person referred to in Section 10.03 k. for the purpose of all the foregoing provisions of this Section 10.04 and for the purposes of Section 10.05, the University is or shall be deemed to be acting as agent or trustee on behalf of and for the benefit of each person who is or who becomes such agent, employee or other person from time to time.
- 10.05 The insurance policies referred to in Section 10.03 may contain such deductible as is determined upon by the Tenant and agreed to by the University, acting reasonably, provided always however that to the extent such policy is not a policy of public liability insurance and contains a deductible, the Tenant shall act as a self-insured and waives recourse co-extensive with the waiver of subrogation referred to in Sections 10.03 and 10.04.

- **10.06** A current certificate(s) evidencing such insurance as required to be carried by it pursuant to the provisions of Section 10.03 shall be provided by the Tenant to the University.
- **10.07** The Tenant shall comply with all lawful requirements of any insurance carrier of the University affecting the Building.
- 10.08 The Tenant shall at its own cost comply with all laws, regulations and government orders or directions relating to the use, generation, manufacture, production, processing, storage, transportation, handling, release, disposal or clean-up of hazardous substances and the protection of the environment on, under or about the Premises. The term "hazardous substances" includes, without limitation, flammables, explosives, radioactive materials, hazardous chemicals or biologicals and waste or substances, petroleum and petroleum products and all substances, materials, goods or gases declared or listed to be hazardous or toxic under laws or regulations in force in the Province. The Tenant shall not use or cause or permit to occur the generation, manufacture, production, processing, storage, handling, release, disposal or clean-up of, any hazardous substance on, under or about the Premises or the transportation to or from the Premises of any hazardous substances except as specifically disclosed to and approved by the University and except in accordance with such requirements of the University as are applicable thereto. Upon the demand of any governmental authority, or the University, that a clean-up be undertaken because of any deposit, emission, leak, spill, discharge or other release of hazardous substance that occurs during the Term on, under, about, at or from the Premises the Tenant shall at its own expense take all remedial action necessary to carry out a full and complete clean-up and shall thereafter provide to the University an independent report or audit in form and substance and from qualified experts approved by the University acting reasonably, certifying such clean-up has been properly completed.
- 10.09 The Tenant shall not do or permit to be done, or omit or permit to be omitted, anything which causes or has the effect of causing the rate of insurance upon the Building to be increased and if the insurance rate shall be increased by any action or inaction of the Tenant or any sub-tenant, licensee or occupant of the Premises permitted by the Tenant, the Tenant shall pay to the University on demand the amount by which the insurance premiums shall be so increased.

Assignment, Subletting, Licensing

- 11.01 The Tenant shall not assign this Lease without the prior written consent of the University first had and obtained, which consent may be withheld for any reason which the University in its sole discretion deems appropriate.
- 11.02 Except as otherwise provided in this Section 11.02 the Tenant shall not sublet all or any part of the Premises, part with possession of all or any part of the Premises or grant a license for all or any part of the Premises without the prior written consent of the University first had and obtained, which consent may be withheld for any reason which the University in its sole discretion deems appropriate. Notwithstanding the foregoing portion of this Section 11.02 the Tenant has the right to enter into a commercial sub-lease of a portion or portions of the Premises with a sub-tenant for

the provision of support services to students as defined in the regulation to the Post-Secondary Learning Act (Alberta) as amended from time to time with the prior written consent of the University not to be unreasonably withheld or delayed provided that:

- a. prior to the consent of the University being effective, the Tenant shall deliver to the University an agreement, in form and content required by the University acting reasonably, in which the proposed assignee or sub-tenant agrees directly with the University to perform and observe all provisions of this Lease applicable to the Tenant;
- **b.** if so required by the University, the form of the commercial sub-lease or license is approved in advance by the University; and
- c. neither the consent by the University pursuant to Section 11.02(a), the approval of the University to the form of commercial sub-lease pursuant to Section 11.02(b), the sub-letting nor the taking by the University of any documentation in relation thereto affects the obligations of the party named as the Tenant in this Lease to perform and observe all provisions of this Lease applicable to the Tenant.

Signs

12.01 The Tenant shall place no signs on the exterior of the Premises or to the extent the same are visible from the exterior of the Premises on the interior thereof or within the interior common areas without the prior written consent of the University which may be arbitrarily withheld.

Damage, Destruction and Expropriation

- **13.01** If during the term hereof the Building shall be damaged or destroyed by fire or any other casualty and:
 - a. subject always to the operation of any applicable terms of the PAW Centre Project Agreement including, without restriction, Section 12.01 to 12.05, the University elects to demolish and not to rebuild the Building, the Term shall cease and be at an end to all intents and purposes effective as of the date of such termination as provided within the PAW Centre Project Agreement; or
 - the University elects to rebuild the Building, this Lease and the Term shall continue, but the Tenant shall not during the period of rebuilding by the University contemplated herein be entitled to use any portion of the Premises which the University deems unsafe or unusable or which interferes with the rebuilding by the University in accordance with the PAW Centre Project Agreement. In this situation, the University shall proceed forthwith and diligently to effect any necessary repairs and rebuilding to the Building and the Premises provided that the University shall not be responsible for the rebuilding or restoration of any Tenant Improvements or Tenant Property.

The University may only elect to demolish and not rebuild the PAW Centre, in accordance with the PAW Centre Agreement.

Quiet Enjoyment

14.01 The University has good right and full power to lease the Premises and to grant the rights and privileges herein set out and so long as the Tenant complies with and performs its obligations under this Lease, it shall quietly enjoy the Premises without hindrance or molestation by the University or any person claiming by, through or under the University, on the terms and conditions set forth in this Lease.

Indemnities

- 15.01 The Tenant shall indemnify and save harmless the University, its officers, agents and employees from any and all liabilities, claims, damages, actions, suits and costs whatsoever brought by any person against the University caused by or arising out of, directly or indirectly:
 - **a.** the performance of this Lease by the Tenant;
 - **b.** the Tenant's occupancy or use of the Premises;
 - **c.** any business or operations of the Tenant or any sub-tenant or licensee conducted in the Premises; or
 - d. by reason of any matter or thing done, permitted or omitted to be done by the Tenant, its agents, officers or employees whether occasioned by negligence or otherwise on, in or in relation to the Premises during the term of this Lease or any renewal thereof.
- **15.02** The Tenant's obligation pursuant to Section 15.01 shall extend to all claims and demands by reason of improper or faulty erection or construction of facilities, trade fixtures or equipment installed on or in the Premises by or on behalf of the Tenant.
- **15.03** The obligation of the Tenant pursuant to Section 15.01 and 15.02 shall survive any termination of this Lease.

Common Use Areas

16.01 The Tenant, its invitees, employees and agents shall have the privilege of utilizing in common with other persons entitled thereto at all reasonable hours, the Common Use Areas in common with others entitled thereto for the purposes designated therefore by the University. The delivery and shipping of merchandise, supplies, fixtures and other materials or goods of whatsoever nature to or from the Premises and all loading, unloading and handling there of shall be done through such entrances and corridors as designated by the University and at such times and by such means as approved by the University.

Exclusion of Liability

- 17.01 Notwithstanding anything in this Lease otherwise contained, notwithstanding that the same may be caused or occasioned by the negligence to any degree of the University, its servants, agents or employees, the University:
 - a. shall not except to the extent the same is recoverable from insurance required to be carried by the University hereunder be liable for damage or injury to any property of the Tenant which is entrusted to the care or control of the University, its agents, servants or employees;
 - b. shall not except to the extent the same is recoverable from insurance required to be carried by the University hereunder be liable or responsible in any way for any personal injury of any nature whatsoever that may be suffered or sustained by the Tenant or any employee, agent, servant or invitee of the Tenant or any other person who may be upon the Premises or the Building or for any loss of or damage or injury to any property belonging to the Tenant or to its employees, agents, servants or invitees or any other person while such property is on or within the Premises or the Building.
- 17.02 Notwithstanding anything in this Lease otherwise contained, neither the University, its agents, servants nor employees shall be liable or responsible in any way for any indirect or consequential losses or damages of any nature whatsoever that may be suffered or sustained by the Tenant or any employee, agent, servant or invitee of the Tenant or any other person who may be upon the Premises howsoever the same are caused, save and except for to the extent caused or contributed to by a default on the part of the University under this Lease or by the negligence of the University, its employees, agents or servants.
- 17.03 Notwithstanding anything in this Lease otherwise contained, the University shall not be liable for any damage or damages of any nature whatsoever and howsoever caused, including by the negligence to any degree of the University, its employees, agents or servants and arising by failure to supply adequate drainage, snow or ice removal, or by reason of the interruption of any public utility or of any service required to be provided to the University or to be provided by the University or in the event that steam, water, rain or snow may leak into, issue or flow from any part of the Building or from the water, steam, sprinkler or drainage pipes or plumbing works or from any other place or quarter or for any damage caused by anything done or omitted by any tenant, but the University shall, after notice of the same and where it is within its obligation so to do, use all reasonable diligence to remedy such condition, failure or interruption of service when not directly or indirectly attributable to the Tenant.

No Liens

18.01 The Tenant shall not suffer nor permit during the term of this Lease any builders' liens or other liens for work, labour, services or material ordered by or at the request of the Tenant or for the

cost of which the Tenant may in any way be obligated, to attach to the title to the Building or the lands on which the same is located or any portion thereof. Whenever and so often, if ever, as any such lien or liens shall be filed or shall so attach, the Tenant will, within thirty five (35) days thereafter, either pay the same or procure the discharge thereof by giving security therefore or in such other manner as is or may be required or permitted by law in order that such lien or liens be removed.

Default

19.01 A Default will occur whenever:

- a. the interest of the Tenant in the Term shall be taken in execution or attachment, which breach is not cured within 30 Business Days after receipt by the Tenant of a notice of such breach from the University (or if the breach reasonably requires more than 30 Business Days to Cure, unless the Tenant commences actions to cure the breach within such period and thereafter promptly, effectively and continuously works to and does remedy and cure the breach); or
- b. the Students' Union and the GSA make a general assignment for the benefit of its creditors or a proposal under the Bankruptcy and Insolvency Act or is adjudged bankrupt or insolvent, or if a receiver shall be appointed for the business, property, affairs or revenues either the Students' Union or the GSA; or
- c. the Tenant shall abandon the Premises, which breach is not cured within 30 Business Days after receipt by the Tenant of a notice of such breach from the University (or if the breach reasonably requires more than 30 Business Days to Cure, unless the Tenant commences actions to cure the breach within such period and thereafter promptly, effectively and continuously works to and does remedy and cure the breach); or
- d. the Tenant shall, or shall purport to, assign, sub-let or grant a license with respect to any portion or all of the Term or the Premises other than in accordance with the terms of this Lease, which breach is not cured within 30 Business Days after receipt by the Tenant of a notice of such breach from the University (or if the breach reasonably requires more than 30 Business Days to Cure, unless the Tenant commences actions to cure the breach within such period and thereafter promptly, effectively and continuously works to and does remedy and cure the breach); or
- e. the Tenant shall not observe, perform and keep any other of the covenants, agreements, provisions, stipulations and conditions herein to be observed, performed and kept by the Tenant which breach is not cured within 30 Business Days after receipt by the Tenant of a notice of such breach from the University (or if the breach reasonably requires more than 30 Business Days to Cure, unless the Tenant commences actions to cure the breach within such period and thereafter promptly, effectively and continuously works to and does remedy and cure the breach);

f. the University shall not observe, perform and keep any other of the covenants, agreements, provisions, stipulations and conditions herein to be observed, performed and kept by the , which breach is not cured within 60 Business Days after receipt by the University of a notice of such breach from the Tenant (or if the breach reasonably requires more than 60 Business Days to Cure, unless the University commences actions to cure the breach within such period and thereafter promptly, effectively and continuously works to and does remedy and cure the breach).

The Tenant or University shall have the right to deliver to the other Party a Notice specifying the event of Default in respect of the defaulting Party. The Party receiving the Default Notice shall have a period of twenty (20) Business Days from the date of its receipt of the Notice within which to deliver a Notice to the other party that it contests the existence of a default (the "Objection Notice"). Failure to deliver an Objection Notice within the specified time shall be deemed an acknowledgement that the Default has occurred.

If an Objection Notice is delivered, the determination as to whether a Default exists shall be referred to arbitration pursuant to the provisions of Section 19.08 hereof. If an Objection Notice is delivered, the curative provision set out in the definition and in Section 19.02, if any, shall begin on the date of the arbitrator's determination that the Default existed.

- 19.02 If the defaulting Party has not cured, or where otherwise permitted pursuant to Section 19.01 commenced and diligently pursued cure of, the Default specified in the Notice referred to in Section 19.01 within the time periods provided for in this Agreement:
 - a. the other Party shall have the right:
 - i. to bring any proceedings in the nature of specific performance, injunction or other equitable remedy, it being acknowledged by the parties that damages at law may be an inadequate remedy for a default, breach or threatened breach of this Agreement; and
 - ii. to bring any action at law or in equity as may be permitted in order to recover damages or for such other remedy or remedies as may be available to it; and
 - and the defaulting Party is the Tenant, the University may re-enter upon and take possession of the Premises or any part thereof in the name of the whole, without limiting any of its other rights at law, and if and whenever the University is entitled to re-enter the Premises, the University may terminate this Lease and the Term by giving written notice of termination to the Tenant and in such event the Tenant will forthwith vacate and surrender the Premises
- 19.03 The failure of the University to insist in any one or more cases upon the strict performance of any of the covenants of this Lease or to exercise any option herein contained shall not be construed as

a waiver or a relinquishment for the future of such covenant or option and the acceptance of Rent by the University with knowledge of the breach by the Tenant of any covenants or conditions of this Lease shall not be deemed to be a waiver of such breach and no waiver by the University of any provisions of this Lease shall be deemed to have been made unless expressed in writing and signed by the University.

- **19.04** If at any time and so often as the same shall happen:
 - a. the Tenant shall be in Default as set forth within Section 19.01 of this Lease; or
 - b. the Tenant shall not observe, perform and keep any of the covenants, agreements, provisions, stipulations and conditions herein to be observed, performed and kept by the Tenant and the Landlord, acting reasonably, believes that such failure on the part of the Tenant results in a risk of material damage, loss or injury to the Building or any portion thereof or persons therein or to the University;

then the University may, but shall not be obligated to, without waiving or releasing the Tenant from its obligations under the terms of this Lease, itself observe and perform the covenant or covenants in respect of which the Tenant is in default, and in that connection may pay such monies as may be required or as the University may reasonably deem expedient, and the University may thereupon charge all monies so paid and expended by it to the Tenant together with interest thereon from the date upon which the University shall have paid out the same; provided however that if the University commences and completes either the performance of any such covenant or covenants or any part thereof, the University shall not be obliged to complete such performance or be later obliged to act in like fashion.

- 19.05 Unless otherwise expressly provided in this Lease, all sums and costs paid by the University, including costs paid between solicitor and client, on account of any default by the Tenant under this Lease shall be payable to the University by the Tenant forthwith, with interest thereon at the rate hereinbefore set forth from date of payment of such sums or costs by the University.
- 19.06 All rights and remedies of the University in this Lease contained shall be cumulative and not alternative and are not dependent the one on the other and mention of any particular remedy or remedies of the University in respect of any default by the Tenant shall not preclude the University from any other remedy in respect thereof, whether available at law or in equity or as expressly provided for herein.
- 19.07 In the event the Tenant remains in possession of the Premised after the end of the Term and without the execution and delivery of a new Lease, there shall be no tacit renewal of this Lease and Term hereby granted and the Tenant shall be deemed to be occupying the Leased Premises as a Tenant from month to month on the terms and conditions contained herein; provided however that this provision shall not authorize the Tenant to so over hold where the University has objected to such over holding or has required the Tenant to vacate the Premises.

- **19.08** If an Objection Notice is delivered pursuant to Section 19.01 hereof, the following provisions shall apply:
 - a. the Parties shall have a period of fifteen (15) days following receipt of the Objection Notice to agree upon a single arbitrator. If within such period of fifteen (15) days, the Parties cannot agree on a single arbitrator, each Party shall within a further period of fifteen (15) days select and appoint an arbitrator. The two appointed arbitrators shall, within fifteen (15) days of the appointment of the second arbitrator, appoint a third arbitrator;
 - **b.** if either Party fails to appoint an arbitrator within the said fifteen (15) day the arbitrator appointed by the other Party shall determine the matter and the arbitrator's decision shall be final and binding upon the Parties;
 - c. if the first two arbitrators appointed in accordance with this Section fail to appoint a third arbitrator within the said fifteen (15) day period following the appointment of the second arbitrator, the third arbitrator shall be appointed by a Justice of the Court of Queen's Bench of Alberta on the application made by either Party to that Court;
 - **d.** the arbitrator(s) shall be requested to make a decision relating to the matter in question within thirty (30) days of the determination of the panel of arbitrators;
 - e. each Party shall, in the case of a panel of three arbitrators, pay the expenses and fees of its arbitrator, and in addition, shall pay one half of the fees and expenses of the third arbitrator; and in the case of a single arbitrator, each Party shall pay one half of the fees and expenses of that arbitrator; and
 - **f.** except as otherwise provided, the provisions of the Arbitration Act of the Province of Alberta or any similar or substitute Act then in force shall apply.

Notices

20.01 Any demand or notice to be given pursuant to this Lease, unless otherwise expressly provided herein, will be duly and properly made and given if made in writing and delivered to the address of the party for whom it is intended as set out below in a cover addressed to such party as follows:

a.	in the case of the Tenant:
b.	in the case of the University:

or to such other addresses as the parties may from time to time designate in writing, and any demand or notice so made or given will be deemed to have been duly and properly made or given and received on the day on which it shall have been so delivered.

Interpretation

21.01 The headings used throughout this Lease are inserted for reference purposes only and are not considered or taken into account in construing the terms and provisions of any paragraph or Section and are not to be deemed in any way to qualify, modify, or explain the effects of such provisions or terms.

Successors and Assigns

22.01 This Lease shall enure to the benefit of and be binding upon the parties hereto and their respective successors and assigns or where required herein their respective approved successors and assigns.

Time of Essence

23.01 Time is of the essence of this Lease.

Unavoidable Delays

24.01 In the event that the University shall be delayed, hindered or prevented from the performance of any covenant hereunder by any cause not within the control of the University (excluding lack of finance of the University), the performance of such covenant shall be excused for the period during which such performance is rendered impossible and the time for performance thereof shall be extended accordingly, but this shall not excuse the Tenant from the prompt payment of any amount required to be paid by the Tenant under the provisions of this Lease.

Joint and Several Obligations

25.01 All warranties, representations, obligations and liabilities of the Tenant contained in or arising pursuant to this Lease are joint and several warranties, representations, obligations and liabilities, as the case may be, of the Students' Union and the GSA.

IN WITNESS WHEREOF the duly authorized officers of the parties have executed and delivered this Agreement on the dates indicated below.

THE GOVERNORS OF THE UNIVERSITY OF ALBERTA		
	Date:	
Dr. Indira Samaraskekera President		-
	Date:	
Don Hickey Vice-President (Facilities and Operations)		-
THE STUDENTS' UNION OF THE UNIVERSITY OF ALBERTA	Date:	
President		_
THE GRADUATE STUDENTS' ASSOCIATION OF THE UNIVERSITY OF ALBERTA		
	Date:	
President		_

SCHEDULE E-1 COMMITTEE PROCEDURES

ARTICLE 1 - DESIGNATION AND AMENDMENT OF PROCEDURES

- 1.1 The procedures set forth herein shall apply to the operations of the Steering Committee and the Strategic Operating Committee established under the main body of the PAW Centre Agreement.
- 1.2 The Committee may exercise all powers and do all acts and things as the Committee may exercise and do as contemplated within the PAW Centre Agreement, including these Procedures.
- 1.3 Save and except for the quorum requirements contemplated within the main body of the PAW Centre Agreement and Section 6.1 of these Procedures, and the meeting provisions contained within Article 4 of these Procedures:
 - (a) the Members of the Steering Committee may from time to time amend these Procedures, as they relate to the Steering Committee, by passing a Special Resolution to that effect; and
 - (b) the Members of the Strategic Operating Committee may from time to time amend these Procedures, as they relate to the Strategic Operating Committee, by passing a Resolution to that effect.

ARTICLE 2 - INTERPRETATION

- 2.1 In these Procedures, so far as the context does not otherwise require:
 - (a) "Business Day" means a day other than a Saturday, Sunday or statutory holiday in the Province of Alberta;
 - (b) "Chair" means that Committee Member appointed as the Chair under the main body of the PAW Centre Agreement or these Procedures;
 - (c) "Committee" means either the Steering Committee and the Strategic Operating Committee, as the case may be or as the context so requires;
 - (d) "PAW Centre Agreement" means the agreement to which this Schedule E-1 is attached, and of which this Schedule E-1 is a part;
 - (e) "Procedures" means the Procedures of the Committee, as described in this Schedule E-1;

(f) "Members" means the members of the Committee appointed from time to time and a "Member" means one of them;

(g) "Resolution" means:

- (i) a resolution passed at a meeting of the Committee by a simple majority of the votes cast by those Members entitled to vote and present in person or by proxy at the meeting, provided that in the case of the Strategic Operating Committee, the President (or designate) of the Graduate Students' Association, the President (or designate) of the Students' Union, and one of the Members listed in section 6.04(a)-(e) of the PAW Centre Agreement must vote in favour of such resolution; or
- a resolution that has been submitted to the Members and consented to in writing by no less than a simple majority of the Members, in person or by proxy, who would have been entitled to vote on it at a meeting at the Committee, provided that in the case of the Strategic Operating Committee, the President (or designate) of the Graduate Students' Association, the President (or designate) of the Students' Union, and one of the Members listed in s. 6.04(a)-(e) of the PAW Centre Agreement must vote in favour of such resolution;
- (h) "Secretary" means the person appointed as the Secretary under these Procedures; and
- (i) "Special Resolution" means:
 - (i) a resolution passed at a meeting of the Committee by a majority consisting of no less than three-quarters (3/4) of the Members entitled to vote at the meeting; or
 - (ii) a resolution that has been submitted to the Members and consented to in writing by no less than three-quarters (3/4) of the Members, in person or by proxy, who would have been entitled to vote on it at a meeting at the Committee.

Any capitalized terms used within these Procedures which are not specifically defined herein shall have the meaning ascribed to it within the PAW Centre Agreement.

- 2.2 The objects of the Committee are those named in the main body of the PAW Centre Agreement.
- 2.3 Words importing the singular number or the masculine gender shall include the plural number or the feminine gender, as the case may be, and vice versa, and references to persons, organizations or corporations shall include individuals, partnerships, firms, societies and corporations, all as the context requires.

ARTICLE 3 - MEMBERS

- 3.1 Each Member of the Committee shall be appointed in accordance with the terms of the main body of the PAW Centre Agreement.
- 3.2 No act or proceeding of the Committee is invalid by reason of any defect or irregularity in the appointment of a Member.
- 3.3 A person shall immediately cease to be a Committee Member of the Committee:
 - (a) upon giving his resignation in writing to the Committee;
 - **(b)** upon his death; or
 - upon the Party appointing the Member delivering written notice to the Committee that the appointing Party desires the Member to be removed from the Committee.

ARTICLE 4 - MEETINGS OF MEMBERS

- 4.1 Subject always to and in accordance with the main body of this Agreement, the meetings of the Committee may be held at any time and place to be determined by the Chair in consultation with the Committee. Notwithstanding anything in these Procedures to the contrary, the Committee may set a schedule of Committee meetings for a certain period or periods together with the accompanying standing agenda of the business to be transacted:
 - in the case of the Steering Committee, by passing a Special Resolution to that effect; and
 - (b) in the case of the Strategic Operating Committee, by passing a Resolution to that effect:

and, in such case, written notice of each such meeting is not required to be given to the Members. Nothing within this Section 4.1 shall be deemed to waive or amend the requirement to provide notice in writing in accordance with these Procedures of any of the business to be transacted which is not otherwise contemplated within the standing agenda(s) approved by the Committee for such schedule meeting(s).

- **4.2** Subject to Section 4.5, the Committee may, whenever it thinks fit, convene any other meeting of the Committee.
- **4.3** Except as otherwise stated in the Procedures, written notice of:
 - (a) the time and place of meetings of the Members; and
 - (b) an agenda of the business to be transacted;

shall be given to each Member, as applicable, at least seven (7) days before the holding of a meeting. Notice of all such meetings, together with any agenda(s), shall also be provided in writing to the executive directors of each of the Associations concurrently with the provision of such notice and/or agenda(s) to the Members of any Committee.

- **4.4** Subject to the provisions of these Procedures, meetings of Members may be called at any time by the Chair.
- 4.5 A meeting of Members shall be called by the Chair or Secretary upon receipt of a petition signed by no less than one-third of the Members setting forth the reasons for calling such meeting.

ARTICLE 5 - ERROR OR OMISSION IN NOTICE

- 5.1 No error or omission in giving notice of any meeting or any adjourned meeting or the associated agenda shall invalidate such meeting or make void any proceedings taken thereat provided always that:
 - (a) there are sufficient Members present to constitute a quorum as contemplated under the provisions of the main body of this Agreement and these Procedures; and
 - (b) the waiver of the requirement for notice in accordance with the provisions of these Procedures, and the ratification, approval and confirmation of any or all proceedings taken or had thereat:
 - (i) in the case of the Steering Committee, is consented to by passing a Special Resolution to that effect; and
 - (ii) in the case of the Strategic Operating Committee, in consented to by passing a Resolution to that effect.

ARTICLE 6 - QUORUM

6.1 A quorum for the transaction of business at any meeting of Members shall consist of the respective quorums set forth within the main body of the PAW Centre Agreement for either the Steering Committee or the Strategic Operating Committee.

ARTICLE 7 - PROCEEDINGS AT MEMBERS' MEETINGS

- 7.1 Unless otherwise provided for within these Procedures, neither:
 - the agenda for a meeting of Members that has been provided to Members pursuant to Section 4.3 of these Procedures: nor

(b) the standing agenda of a schedule meeting of Members approved by the Committee pursuant to Section 4.1 of these Procedures;

may be amended at such meeting so as to contemplate any other or additional business to be transacted, unless notice of the revised agenda has been provided to the Members in accordance with Section 4.3 of these Procedures, or such notice of the revised agenda has otherwise been waived in accordance with these Procedures.

- **7.2** Each Member shall be entitled to one (1) vote on every matter properly put before a meeting for a vote.
- 7.3 Every action or decision taken or made by a Committee shall be taken or made by way of a Resolution or Special Resolution, as these Procedures may require.
- Every question shall be decided in the first instance by a show of hands unless any Member demands a recorded vote. Upon a show of hands, every Member shall have one (1) vote, and unless a recorded vote be demanded, a declaration by the Chair that a resolution has been carried or not carried and an entry to that effect in the minutes of the Committee shall be sufficient evidence of the fact without proof of the number or proportion of the vote accorded in favour of or against such resolution. The demand for a recorded vote may be withdrawn, but if a recorded vote be demanded and not withdrawn, the Members present in person or by designate shall vote, and such recorded vote shall be taken in such manner as the Chair shall direct and the result of such recorded vote shall be deemed the decision of the Committee upon the matter in question. In the event of a tie vote, the Chair shall not have a second and casting vote and the motion shall be declared lost.
- 7.5 Each Member may vote by way of a designate. Such designate needs not himself or herself to be a Member.
- 7.6 Business at all meetings of the Committee shall be conducted according to these Procedures and the rules of order adopted by the chair of the meeting, provided always that:
 - (a) such rules shall provide each Member with a reasonable right to due process and right to be heard, subject to a reasonable need to carry meeting efficiently and effectively; and
 - (b) in the absence of rules so adopted, or to the extent that such rules or these Procedures do not provide, Roberts' Rules of Order or respective portions thereof shall govern conduct of the meetings.
- 7.7 A proposed resolution may be submitted to any meeting by any person present and entitled to vote, whether or not notice in writing of the intention to submit such resolution had been given to the Secretary prior to such meeting.

- 7.8 The Chair shall preside at every meeting of the Members but if the Chair is not present, or for any reason the Chair refuses to preside, then those present at the meeting shall appoint a Chair to act at that meeting.
- 7.9 The Chair may move or propose a resolution.

7.10 Where a Member believes a personal conflict of interest on a given matter exists, that Member shall make that declaration at a meeting. If, after making such a declaration, a Member votes in respect of a Resolution or Special Resolution, then the Chair shall pass that declaration along to the appropriate decision-makers within the University, together with the recommendation that is the subject matter of the Resolution. For clarification, no Member shall be deemed to be in a conflict of interest due solely to that Member's past, concurrent, or future anticipated role or position with or within the University, the Board of Governors, or either of the Associations.

ARTICLE 8 - ADMINISTRATION

8.1 A Secretary shall be appointed by the Chair from the Chair's staff, and failing such appointment be elected by the Committee, but does not have to be a Member. If the Secretary is a Member and ceases to be a Member, then such Secretary also ceases to hold office.

8.2 Chair

The Chair shall be ex-officio a member of all Committees. The Chair when present, shall preside at all meetings of the Committee. In the Chair's absence, a chairperson may be elected by those attending the meeting to preside thereat.

8.3 <u>Secretary</u>

- (a) It shall be the duty of the Secretary to attend all meetings of the Committee, to keep accurate minutes or the same, and maintain proper records of the Committee in accordance with University procedures relating to creation, retention and disposition of records. In the absence of the Secretary, the Secretary's duties shall be discharged by such Committee Member as may be appointed by the Committee. The Secretary shall have charge of all the correspondence of the Committee and be under the direction of the Chair and the Committee.
- (b) The Secretary shall also keep a record of all the Members and their respective contact information for notices and agendas, and send all notices and agendas of the various meetings as required.
- (c) The duties of the Secretary may be delegated to such other Party or Parties as the Committee may decide.
- **8.4** The Secretary may be removed from office by a Resolution passed at a meeting of the Committee.

- 8.5 The Committee may appoint, remove and replace the Secretary as it deems necessary and determine the responsibilities and term, if any, of such other officers.
- 8.6 No Secretary shall be remunerated for being or acting in such appointed position, but a Secretary can be reimbursed for all expenses necessarily and reasonably incurred while engaged in the affairs of the Committee.

ARTICLE 9 - SUBCOMMITTEES

- 9.1 The Committee may appoint standing or *ad hoc* subcommittees that it deems necessary consisting of Members and others and may assign duties and responsibilities to the subcommittees that are not inconsistent with the PAW Center Agreement and may make regulations governing their conduct.
- 9.2 At least one (1) Committee Member shall be a member of each subcommittee.
- **9.3** All such subcommittees shall be accountable to the Committee.

ARTICLE 10 - MINUTES

- 10.1 The Secretary will keep all official copies of the approved minutes, which must summarize the discussion at the meetings and which must include:
 - (a) all appointments of the Secretary;
 - (b) all names of all Members present at each meeting of the Members; and
 - (c) all resolutions and proceedings of meetings of the Members and of the Committee.
- Any minutes of the Members and the Committee, if signed by the Chair of that meeting or the Chair of the next succeeding meeting, shall be receivable as prima facie evidence of the matters stated in such minutes

ARTICLE 11 - NOTICE

- 11.1 Any notice which is required or permitted to be given under these Procedures to a Member shall be in writing and may be:
 - (a) delivered personally to the business address on the University campus of the Member to whom it is intended, last known to the Committee as recorded on its books or records;
 - (b) sent by fax to the fax number of the Member last known to the Committee as recorded on its books or records; or

(c) sent by electronic mail to the electronic mail address of the Member last known to the Committee as recorded on its books or records;

and in each case shall be copied to the executive director of each of the Associations.

- Any notice which is required or permitted to be given to the Committee shall be in writing and shall be delivered to the Chair, with a copy of such notice also delivered to the Secretary, in the manners set forth above.
- Any notice delivered before 4:30 p.m. (local time) on a Business Day shall be deemed to have been received on the date of delivery. Any notice delivered after 4:30 p.m. (local time) on a Business Day or delivered on a day other than a Business Day, as the case may be, shall be deemed to have been received on the next Business Day. Any notice sent by fax before 4:30 p.m. (local time) on a Business Day shall be deemed to have been received when the sender receives the confirmation through its fax machine confirming receipt by the recipient; provided however, that any fax received after 4:30 p.m. (local time) on a Business Day, or received on a day other than a Business Day, as the case may be, shall be deemed to have been received on the next Business Day. Any notice sent by electronic mail before 4:30 p.m. (local time) on a Business Day shall be deemed to have been received on the day of sending such notice; provided however, that any electronic mail notice received after 4:30 p.m. (local time) on a Business Day, or received on a day other than a Business Day, as the case may be, shall be deemed to have been received on the next Business Day.

ARTICLE 12 - INSPECTION OF RECORDS BY MEMBERS

12.1 Any Member may inspect the records of the Committee at any time upon giving reasonable written notice to the Committee and arranging a time satisfactory to the Chair or Secretary, respectively. Each Member shall, at all times, have access to such records.

ARTICLE 13 - DISSOLUTION OF COMMITTEE

- 13.1 The Committee shall automatically dissolve upon:
 - (a) in the case of the Steering Committee, upon the Acceptance Date; and

in the case of the Strategic Operating Committee, upon the termination of the PAW Centre Agreement.

SCHEDULE F-1

CONFLICT AND DISPUTE RESOLUTION PROCEDURE

- **1. Definitions** In this Schedule, the following words and phrases have the following meanings:
 - (a) "Arbitrator" means the person appointed to act as such to resolve any Dispute;
 - (b) "Arbitration" means a process whereby each of the Parties, with or without legal counsel, jointly engage and meet with an Arbitrator who will render a binding decision in respect of any Disputes;
 - (c) "Conflict" has the meaning set out in s. 16.01 of the main body of this Agreement;
 - **(d)** "**Disclosed Information**" means the information disclosed by a Party for the purpose of settlement, Negotiation, Mediation or Arbitration;
 - (e) "Dispute" has the meaning set out in s. 16.01 of the main body of this Agreement;
 - (f) "Mediation" means a process whereby a Representative of each Party, with or without legal counsel, agrees to jointly engage the services and meet with a Mediator to participate in a mediation, conciliation or similar dispute resolution process;
 - **(g)** "**Mediator**" means the person appointed to facilitate the resolution of a Dispute between the Parties;
 - (h) "Negotiation" means a process whereby a Representative of each Party, with or without legal counsel, meet to discuss and attempt to resolve a Conflict or a Dispute, at all times acting in good faith;
 - (i) "Party" means a party to this Agreement to which this Conflict and Dispute Resolution Procedure is attached, and "Parties" means more than one of them; and
 - (j) "Provost and Vice-President Academic" means the individual holding the position of Provost and Vice-President Academic from time to time, or its successor position, within the administration of the University of Alberta;
 - (k) "Representative" means
 - (i) in respect of the GSA, a member of the Board of the GSA as may be appointed from time to time;
 - (ii) in respect of the Students' Union, a member of the Students' Council of the Students' Union as may be appointed from time to time;
 - (iii) in respect of the University, the Dean of the Faculty of Physical Education and Recreation or designate, and the Vice-President (Facilities and Operations) or designate, or one of them, as determined by the Dean of the Faculty of Physical Education and Recreation and the Vice-President (Facilities and Operations).
- **2. Conflict Resolution Process** In the event of any Conflict, the Parties agree that the Conflict shall be resolved in the following order:

- (a) first, by Negotiation; and
- **(b)** second, by way of reference of the Conflict to the Provost and Vice-President Academic for resolution.

Negotiation, and if applicable the ultimate resolution by the Provost and Vice-President Academic, shall refer to, take into account, and apply the Shared Mission, Agreement Principles and Project Purpose stated by the Parties within the Agreement to which this Schedule is attached.

- **3. Dispute Resolution Process** In the event of any Dispute, the Parties agree that prior to commencing litigation, they shall undertake a process to promote the resolution of a Dispute in the following order:
 - (a) first, by Negotiation;
 - **(b)** second, by way of Mediation; and
 - (c) third, by arbitration, if mutually agreed to in writing at the time of the Dispute, by the Parties.

Negotiation, Mediation or Arbitration shall refer to, take into account, and apply the intentions and principles stated by the Parties within the Agreement to which this Schedule is attached.

- 4. Negotiation A Party shall give written notice ("Conflict Notice" or "Dispute Notice" as the case may be) to the other Party of a Conflict or Dispute and outline in reasonable detail the relevant information concerning the Conflict or Dispute. As soon as possible following receipt of the Conflict Notice or Dispute Notice, the Representatives shall meet and attempt to resolve the Conflict or Dispute through Negotiation. If a Conflict as described in section 6 of Schedule B-1 or in section 5.09 of this Agreement is not resolved within seven (7) days of receipt of the Conflict Notice, the Negotiation shall be deemed to have failed. If a Conflict as described in section 3.04 or 6.10 of this Agreement, or a Dispute is not resolved within thirty (30) days of receipt of the Conflict Notice or Dispute Notice, the Negotiation shall be deemed to have failed.
- **5. Mediation** If the Representatives cannot resolve a Dispute within such thirty (30) day period, then the Dispute shall be referred to Mediation, in accordance with the National Mediation Rules of the ADR Institute of Canada, Inc. Any one of the Parties shall provide the other Parties with written notice ("**Mediation Notice**") specifying the subject matters remaining in Dispute, and the details of the matters in Dispute that are to be mediated. If the Mediation is not completed within sixty (60) days from the date of receipt of the Mediation Notice, the Dispute shall be deemed to have failed to be resolved by Mediation.
- 6. Conflict Resolution If the Negotiation fails to resolve a Conflict, the Conflict shall be submitted to the Provost and Vice-President Academic for binding resolution. The Provost and Vice-President Academic's decision is final and binding. The Provost and Vice-President Academic shall meet with the Representatives together to hear and render a written decision concerning any Conflict within forty-five (45) days of the termination of Negotiation, and shall refer to, take into account, and apply the Shared Mission, Agreement Principles and Project Purpose stated by the Parties within the Agreement to which this Schedule is attached.

7. Dispute Arbitration

- (a) If the Mediation fails to resolve the Dispute and if both Parties so agree in writing, at the time of the dispute or if so provided in the body of the Agreement of which this schedule forms a part, the Dispute shall be submitted to binding Arbitration. One of the Parties may provide the other Party with written notice ("Arbitration Notice") specifying the subject matters remaining in Dispute and the details of the matters in Dispute that are to be arbitrated. If the other Party agrees to proceed to Arbitration, such Dispute shall proceed to Arbitration. A failure to respond to the Arbitration Notice shall be deemed to constitute a refusal to proceed with Arbitration;
- (b) The Arbitrator shall conduct the Arbitration in accordance with the commercial arbitration rules (the "Rules") established from time to time by the ADR Institute of Canada Inc., unless the Parties agree to modify the same pursuant to any arbitration agreement. The Arbitration Act (Alberta) shall apply to all Arbitrations but if there is a conflict between the Rules and the provisions of the Act, the Rules shall prevail. Notwithstanding the foregoing, any such Arbitration shall be conducted in the English language;
- (c) The Parties shall request that the Arbitrator proceed to hear and render a written decision concerning any Dispute within:
 - (i) forty-five (45) days, if the subject matter of the Dispute is less than \$50,000.00; or
 - (ii) one hundred and twenty (120) days, if the subject matter of the Dispute is greater than \$50,000.00.
- (d) The Arbitrator has the right to award solicitor-client costs against the unsuccessful Party and to award interest but does not have the right to award punitive, consequential or other exemplary damages.
- (e) The Arbitrator's decision is final and binding but is subject to appeal or review by any court of proper jurisdiction only with respect to an allegation of fraud.
- (f) The Arbitrator shall at all times, and in making its decision, have due regard for the Shared Mission, the Agreement Principles and the Project Purpose.
- **8. Participation** The Parties and their Representatives will participate in good faith in the Negotiation, and where and if applicable Mediation, the resolution by the Provost and Vice-President Academic or Arbitration processes and provide such assistance and Disclosed Information as may be reasonably necessary and notwithstanding that litigation may have commenced as contemplated in this Schedule.

- **9. Location** The place for all Negotiation, Mediation, resolution by the Provost and Vice-President Academic, and Arbitration shall be upon the campus of the University of Alberta in Edmonton, Alberta, and at such specific location as the Parties may agree or as the Mediator, the Provost and Vice-President Academic, or the Arbitrator shall direct.
- 10. Selection of Arbitrator If the Parties are unable to agree upon the appointment of a single Arbitrator within ten (10) days after receipt of the Arbitration Notice, either of the Parties may request that a single Arbitrator, of suitable training, experience and independence, and who in respect of the subject matter of the Dispute has a reasonable practical understanding, be appointed by the executive director or other individual fulfilling that role for the ADR Institute of Canada, Inc. The executive director shall be requested to make this determination within five (5) days of receipt of the request.
- 11. Costs Subject to clause 7(d) of this Schedule in the case of an Arbitration, the Parties shall bear their respective costs incurred in connection with the Negotiation, Mediation, and where and if applicable resolution by the Provost and Vice-President Academic or Arbitration, except that the Parties shall equally share the fees and expenses of the Mediator and Arbitrator and the cost of the facilities required for Mediation and Arbitration.
- 12. Disclosed Information All Disclosed Information shall be treated as confidential and neither its delivery nor disclosure shall represent any waiver of privilege by a Party disclosing such Disclosed Information. Subject only to the rules of discovery or other applicable legislation compelling disclosure, each Party agrees not to disclose the Disclosed Information to any other person or for any other purpose. Such Disclosed Information cannot be used in any subsequent proceedings without the consent of the Party who has made the disclosure. The Parties agree that any Representative, Mediator and, if applicable, Arbitrator shall not be subpoenaed or otherwise compelled as a witness in any proceedings for the purpose of testifying with respect to the nature or substance of any dispute resolution process that may arise in relation to any matter that is a subject of this Agreement. Nothing in this dispute resolution procedure shall require a Party to disclose information that is subject to confidentiality provisions with third parties.
- 13. Litigation and Limitations Act No Party shall commence litigation concerning a Dispute until the Negotiation and Mediation processes have concluded. The Parties agree that during the time any Dispute is subject to the Negotiation and Mediation processes, the limitation periods set forth in the Limitations Act (Alberta) shall be stayed. The limitation periods shall be reinstated once the Mediation terminates or is deemed terminated so that each of the Parties shall have the respective rights and remedies that were available to them before the commencement of these processes. Any Party may commence litigation on any date, if necessary, to preserve its legal rights and remedies if the commencement of litigation after that date would otherwise be banned by any applicable limitation period or if the commencement of litigation is otherwise necessary to prevent irreparable harm to that Party.



POLICY COMMITTEE MEETING SUMMARY REPORT TO COUNCIL

Date: January 31, 2012 Time: 6:37 PM

Mot	tions	
1.	WOODS moved that the January 31, 2012 agenda be approved as tabled.	CARRIED 7/0/0
2.	KHINDA moved that the January 17, 2012 minutes be approved as amended.	CARRIED 7/0/0



BUDGET AND FINANCE COMMITTEE MEETING SUMMARY REPORT TO COUNCIL

Date: Feb 15, 2011 Time: 5.30 pm 2011 – 2012 MEETING 10

Mot	ions	
1.	CRONE/LE MOVES to postpone approval of the February 1 minutes until the next meeting of the Budget and Finance Committee.	CARRIED 4/0/0
2.	CRONE/CHEEMA MOVES THAT the Budget and Finance Committee recommend that Students' Council approve the 2012-2013 Budget Principles, as amended.	CARRIED 4/0/0
3.	CHEEMA/LE MOVES to appoint Lyndon Crone as the interim chair of the Budget and Finance Committee from February 17 to March 8.	CARRIED 4/0/0



Office of the VICE PRESIDENT (EXTERNAL)

February 17th, 2012

To: Students' Council

Re: VPX Report to Council

I am doing my report a little early this week since I will be on a leave of absence until March 9th.

Lobby Training

It's happened! On February 11th the Students' Union hoste a pilot project for Advocacy Training. About 20 students came, on such short notice. I believe it was beneficial to everyone who attended, and I already have some ideas on how to improve it next year if the incoming VP (External) is interested.

A special thanks goes to all the people who participated:, Amanda Henry (Student Governance Advisor), Justin Williams (Director of Research and Political Affairs), and Emerson Csorba.

Emerson was interested in creating Lobby Training 2.0 for the incoming exec, council and COFA in April. Maybe we could start doing the tweaks to it then.

Changes to Student Financial Assistance

The Minister of Advanced Education and Technology has announced on Monday February 13th major changes to the Alberta Student Loans Program. Changes to the program include the introduction of a flat-rate contribution of \$1,500 (reduced to zero for single parents), and the elimination of savings, RRSP, part-time earnings or parental contributions in determining loan eligibility.

They will now also stop collecting interest on the loan during the grace period. Moreover Alberta Advanced Education and Technology decided to cut it's Remission Program, and instead move the \$69 millions dollars used to fund this program to a completion grant. All students who have student loans will now get a \$2,000 grant upon their graduation.

The SU, and CAUS, have advocated for much of these changes in the past. However, we will continue to push for more grants instead of loans to reduce overall debt-load and encourage post-secondary participation.

CAUS' number one priority remains to be Mandatory Non-Instructional Fees, and we're hoping to see if the Minister will solve the issue before the election. The following link is my letter to the Journal that was published on Sunday 12th: http://www.edmontonjournal.com/opinion/University+fees+have+real/6140479/story.html

Cheers, Farid



2011/2012 Report to Council

February 17, 2012 Vice President (Student Life)

Prepared by: Colten Yamagishi, Vice President (Student Life)

To: Students' Council 2011/12

Hello Council,

I will be on a leave of absence to run in the SU exec elections at the time of this meeting so I will be absent from Council Chambers but I am writing this report to provide information on the Food Policy and Health and Wellness Policy that will be brought up at this meeting.

Programming:

Programming will be working on a number of projects in the following months including the SU Awards Night, two movie nights, and a number of shows in our venues. At this time we have decided not to pursue a large-scale event such as a Campus Musical, or Talent Show due to a lack of time and resources but we are looking into a possible smaller year-end show/activity. I had a meeting with the programming department to summarize our year and I cannot explain how happy I have been with their performance. Every event this year has grown in participation, quality, and general success thanks to their hard work.

Services

Many of the services will soon begin hiring for Director and Associate Director positions for the following year. Make sure to check out Jobkin to see what is available!

A few updates include the following: CFB will be moving to its new upstairs location over the reading week, Orientation is recruiting volunteers, and the SU awards night is happening at the end of March.

Advocacy

International Students' Association:

We had our first meeting and it was a great success! The VPA will be meeting with the International Centre next week to continue discussions on the creation of this great group.

Sustainability:

The UofA has been awarded with a Silver S.T.A.R.S. rating from the AASHE organization for its continued success in promoting sustainability in its operations on campus and in the community. This is an outstanding achievement so please be sure to congratulate the Office of Sustainability and SUSTAIN staff and volunteers!

Food Policy:

This policy relays many of the themes that I promoted in the presentation that I did with Nikki Way from the Sustainable Food Initiative. I would like to inform council that we had ample

discussion regarding the Fair Trade, Local, and Organic clause and even had the council of a number of Fair Trade and organic food experts while developing the policy in first reading.

Health and Wellness Policy:

I am assuming that this policy will foster a large amount of debate at council. A few of the principles that we wanted to receive advice from council included the following.

-Should we specifically mention emerging addiction problems such as gaming/internet addiction, use of study drugs such as dexedrin, online gambling, etc. with a clause such as the following:

the Students' Union advocates that the University provides addiction counseling services for emerging problems

Hopefully there will be good debate! I will try and sneak into council as a guest after the Residence debate.

Opportunities for you to get involved:

Make sure to vote in the SU exec elections and consider running for council!

Colten Yamagishi 780-299-3076

STUDENTS' COUNCIL VOTES AND PROCEEDINGS

Tuesday February 7th, 2012 Council Chambers 2-1 University Hall

VOTES AND PROCEEDINGS (SC 2011-19)

2011-19/1 SPEAKER'S BUSINESS

2011-19/2b made special order to be moved onto main agenda

2011-19/1a Announcements – The next meeting of Students' Council will take place on

Tuesday, February 28, 2012

2011-19/2 PRESENTATIONS

2011-19/2a PAW Centre Agreement- Presented by Marc Dumouchel and Rory Tighe,

President. Sponsored by Rory Tighe, President/Zinyemba.

Abstract:

This Presentation will give the highlights of the PAW Centre Agreement and the process of how we got here. It will specifically highlight our compliance with the

referendum questions passed by students.

2011-19/2b NSSA Presentation- Presented by Brianne Lovstrom. Sponsored by Emerson

Csorba, VP Academic/Roberto-charron.

Here is our abstract:

The NSSA represents all students at the University of Alberta who have their major or minor listed as Native Studies, are in a combined degree program with the Native Studies Faculty, or are enrolled in one Native Studies course for that academic year. We are very proud to have increased our active membership by over double this past year and to announce that we are applying to be released from probation on February 17th. In our presentation we will address our probationary journey, strategic plan for the future, and address the sustainability of our organization.

o. ou. o. je..._e.

Nicol opposed

YAMAGISHI/ ROBERTO CHARRON MOVED TO make item 2011-19/6a a special order to be moved onto the main agenda.

Motion: FAILED

2011-19/3 EXECUTIVE COMMITTEE REPORT

Rory Tighe, President- Oral Report

Emerson Csorba, VP Academic- Oral Report

Colten Yamagishi, VP Student Life- Oral Report

Farid Iskandar, VP External- Oral Report

2011-19/4 BOARD AND COMMITTEE REPORTS

Petros Kusmu, Policy Committee Chair- Oral Report

Saadiq Sumar, GAC Chair- Oral Report

Adam Woods, Bylaw Committee Chair- Oral Report

Su Su Liang, Audit Committee Chair- Oral Report

2011-19/5 QUESTION PERIOD

Question Period extended 15 minutes

2011-19/7 GENERAL ORDERS

2011-19/7a

BELLINGER/CSORBA MOVE TO approve the Nursing Undergraduate Association proposal as amended to create a Faculty Association Membership Fee.

Speakers List: McCutchern, Mac Gillivary, Crone, Csorba, Tighe, Yamagishi, Liang, Mc Gim, Iskandar

ISKANDAR/FERGUSON MOVED TO recess for 5 minutes.

Motion: CARRIED

Recess at 8:40

Meeting called back to order at 8:46

YAMAGISHI CALLED for division

Tighe- Y
Csorba- Y
Iskandar- Y
Cheema- Y
Yamagishi- Y
Lepage Fortin- Y
Fehr- Y
McGim- Y
Khinda- Y
Woods- Y
Kelly- Y
Mills- Y
Sereda- Y
Le- Y

McMurtrie- Y
Crone- Y
Gould- Y
Bellinger- Y
Zhao- Y
Power- Y
Ferguson- Y
Roberto Charron- Y
Liang- Y
MacGillivery- Y

Motion: CARRIED

Meeting adjourned at 8:55pm

Councillor Attendance R	ecoras				
2011-2012		Attendand	се		
			/ 🐧	81.1/2/05 11.18 05.11.18 01.201.2	/
		/	1	12/	\.\\\
		/	v	/~ /.	~ /
		/ 💰	S' /	S / S	. /
		/ 6/4	/ 6	v / ,v	
O	N.	15	/\v	/ <i>\</i> \`.	/
Council Seats (40 total)		<u> </u>	<u>/o`</u>	/ öV /	
Ex-officio Members (6 voting s	seats)				
Ex-officio Members (6 voting s					
President		V	v	v	
	Rory Tighe	Y	Y	Y	
VP Academic	Emerson Csorba	Y	Y	Y	
VP External	Farid Iskander	Y	Y	0.5	
VP Operations & Finance	Andy Cheema	Y	Y	Y	
VP Student Life	Colten Yamigishi	Y	Y	Y	
Undergraduate Board of Governors Rep	Raphael Lepage Fortin	Y	Υ	Y	
Faculty Representation (32 vo	ting seats)				
ALES	Andrew Fehr	Y	Υ	Y	
Arts	Petros Kusmu	Ÿ	Ÿ	Y(p)	
Arts	Navneet Khinda	Ÿ	Ÿ	Y	
Arts	Adam Woods	0.5(p)	Ÿ	Ÿ	
Arts	Brent Kelly	υ. 3(μ)	Y	Ÿ	
Arts	Chaka Zinyemba	N	Ÿ	0.5	
Arts	Kelsey Mills	Y	Y	V.5	
Arts Augustana (Faculty)	Nathan Sereda	Y	0.5	Y	
Business	Josh Le	Y	0.5 Y	Y	
Business	vacant			-	
Education	Vacant Vanessa Johnson	Y	N	0.5	
Education Education	Brit Luimes	Y	Y	0.5 N	
Education	Mallory McMurtrie	Y	0.5	Y	
Engineering	Aaron Eslinger	Y	0.5	0.5	
Engineering	Lyndon Crone	Y	Y	Y	
Engineering	Saadiq Sumar	0.5	Y	N	
Engineering	Ahmed Elmallah	N	N	N	
Law	Scott Nicol	Y(p)	Υ	N	
Medicine & Dentistry	Rebecca Gould	Υ	Υ	Y	
Native Studies	Vacant				
Nursing	Eric Bellinger	Y	N	Y(p)	
Nursing	vacant				
Open Studies	Vacant				
Pharmacy	Sarah Zhao	Y	Y	Y	
Phys Ed & Rec	vacant				
Saint-Jean (Faculty)	Kinnar Power	Υ	Υ	Y	
Science	Kim Ferguson	Υ	Y	Y	
Science	Amelie Roberto Charron	Y	0.5	Y	
Science	Avinash Karuvelil	0.5	Υ	N	
Science	Peter West	0.5(p)	Y(p)	0.5(p)	
Science	Arun Thomas	Y	0.5	N	
Science	Su Su Liang	Y	Υ	Y	
Science	Brett MacGillivary	Y	0.5	Ÿ	
		-		-	
Ex-Officio Members (2 non-vot	ting seats)				
Speaker	Michael Ross	Y	Υ	Υ	
General Manager	Marc Dumouchel	Y	Y	0.5	
Resigned:					
			I		
	i .	1			